# Limestone Massifs: A Comprehensive Geomorphological and Environmental Analysis

# Introduction

# Defining the Limestone Massif: A Synthesis of Geology and Topography

A limestone massif, a term derived from the French *Le Massif Calcaire*, represents a distinct and globally significant landform defined by the convergence of geology and topography. Geologically, it is a large, coherent, and elevated block of the Earth's crust composed predominantly of carbonate rocks, chiefly limestone and dolostone. Topographically, it is characterized by significant relief, often manifesting as mountain ranges, highlands, or expansive plateaus that stand prominently above the surrounding landscape. However, a limestone massif is more than a static geological feature; it is a dynamic system shaped by a continuous sequence of processes spanning hundreds of millions of years. This report will trace the lifecycle of these massifs, from their genesis as sediments in ancient seas, through their tectonic uplift and subsequent transformation by the unique erosional process of karstification, to their contemporary role as critical ecosystems and complex arenas of human interaction.

# **Global Significance and Thematic Structure**

Carbonate rocks with the potential for karst development underlie approximately 15-20% of the Earth's ice-free continental surface, making limestone massifs a widespread and vital component of the global landscape. Their significance is multifaceted. They are immense geological archives, with their fossiliferous layers providing an unparalleled record of ancient life and organic evolution. They function as crucial hydrological systems, hosting vast karst aquifers that supply potable water to an estimated one billion people worldwide. These landscapes are also biodiversity hotspots, where extreme environmental gradients create a

mosaic of specialized habitats that foster high levels of endemism both on the surface and in extensive subterranean realms. Economically, limestone is an indispensable industrial mineral commodity, fundamental to construction, agriculture, and manufacturing. This report provides a comprehensive analysis of these remarkable landscapes, structured to guide the reader from their fundamental geological origins to their complex environmental and societal roles. Part I examines the genesis of the rock and the tectonic forces that create the massif. Part II details the intricate process of karstification that sculpts the landscape. Part III presents a global tour of prominent massifs, illustrating the diverse expressions of these processes. Part IV explores the unique ecosystems and biodiversity harbored within these environments. Part V analyzes the multifaceted interactions between humans and karst landscapes, from resource extraction to conservation. Finally, Part VI addresses the pressing challenges and strategies for the sustainable management of these invaluable, yet exceptionally fragile, systems.

# Part I: The Genesis of Limestone Massifs

#### **Section 1: From Marine Sediment to Solid Rock**

#### 1.1 Geological and Mineralogical Composition

Geologically, limestone is defined as a sedimentary rock containing 50% or more calcium carbonate (CaCO3), which exists primarily as the mineral calcite.<sup>3</sup> Closely related is dolostone, a rock composed mostly of the mineral dolomite (

CaMg(CO3)2), which forms when magnesium in pore water replaces some of the calcium in the original limestone.<sup>3</sup> In industrial and commercial contexts, the two are often grouped under the general term "limestone" as their uses can be interchangeable.<sup>4</sup> A key diagnostic feature of limestone is its vigorous effervescent reaction with dilute hydrochloric acid, which releases carbon dioxide gas, a property not shared by dolostone unless it is in powdered form.<sup>4</sup> The calcium carbonate in limestone can also exist as aragonite, a polymorph with the same chemical formula as calcite but a different crystal structure. Aragonite is typically of primary biogenic origin, forming the skeletons of modern corals, but it is less stable over geological time and tends to recrystallize into calcite.<sup>3</sup>

#### 1.2 Biogenic and Chemical Formation

The vast majority of limestone deposits of commercial and geological importance were

formed in relatively shallow, warm marine environments.<sup>3</sup> Their origin is predominantly biogenic, resulting from the accumulation of calcium-rich skeletal debris from countless marine organisms.<sup>12</sup> Creatures ranging from microscopic algae (coccoliths) and foraminifera to larger organisms like corals, mollusks, and brachiopods extract dissolved calcium and bicarbonate ions from seawater to construct their protective shells and skeletons.<sup>3</sup> Upon their death, these calcareous remains sink to the seafloor, accumulating over millions of years in thick layers that can cover hundreds of square miles.<sup>3</sup> This makes limestone a profoundly biological rock, a fossiliferous archive that preserves a detailed record of past marine ecosystems.<sup>9</sup>

While biogenic accumulation is the primary pathway, limestone can also form through direct chemical precipitation. This occurs in specific environments where water becomes supersaturated with calcium carbonate. For example, oolitic limestones are composed of small, spherical grains (ooids) formed by the concentric precipitation of calcite around a nucleus in agitated, shallow tropical seas, such as those found today on the Bahama Banks.<sup>17</sup> In continental settings, travertine and tufa form from the rapid degassing of carbon dioxide from calcium-rich water at springs or in caves, causing calcite to precipitate.<sup>3</sup>

#### 1.3 The Process of Lithification

The transformation of loose carbonate sediment—such as shell fragments and calcareous mud-into solid rock is achieved through a process known as lithification. 18 Lithification is a key stage within the broader suite of post-depositional changes referred to as diagenesis. 15 It primarily involves two mechanisms: compaction and cementation. As layers of sediment accumulate, the increasing weight of the overlying material exerts pressure, compacting the lower layers, reducing pore space, and expelling trapped water.<sup>17</sup> Following compaction, cementation occurs as water percolating through the remaining pores dissolves minerals and then reprecipitates them as crystals that grow between the sediment grains. In carbonate sediments, this cement is typically calcite, which effectively glues the grains together, converting the unconsolidated mass into a solid, durable sedimentary rock.<sup>18</sup> The formation of limestone represents a critical component of the Earth's long-term carbon cycle. The initial biogenic process, where marine organisms build their skeletons, actively removes carbon from the oceanic-atmospheric system.<sup>3</sup> As these remains accumulate and undergo lithification, this biologically fixed carbon is sequestered into the solid geosphere, where it can remain locked away for hundreds of millions of years. 15 Consequently, the very existence of the world's great limestone massifs signifies a massive historical withdrawal of atmospheric carbon, directly linking the formation of these geological structures to the planet's long-term climate regulation.

#### Section 2: The Ascent of the Massifs

#### 2.1 The Role of Tectonic Uplift and Orogeny

The thick sequences of limestone formed on ancient sea floors must be raised above sea level to create the highlands and plateaus that define a massif. This elevation is primarily a product of large-scale tectonic forces, particularly those associated with orogeny, or mountain-building events. During orogenies, the collision of continental plates generates immense compressive stress, causing the crust to buckle, thicken, and rise. This process of tectonic uplift can transform a former low-relief plain or shallow seabed into a high-altitude plateau, which is then subjected to erosion. The formation of the Alps and the Pyrenees, for instance, powerfully raised adjacent regions like the Massif Central in France, exposing its ancient granitic and metamorphic core and flanking it with younger limestone plateaus.

#### 2.2 Structural Geology: Folding, Faulting, and Relief

The immense compressive forces of an orogeny do not simply lift the rock layers uniformly; they deform them. The relatively brittle layers of limestone and associated sedimentary rocks respond to this stress by folding and fracturing.<sup>20</sup> This deformation creates large-scale geological structures such as anticlines (upward-arching folds) and synclines (downward-troughing folds), as well as faults, which are fractures in the rock mass along which movement has occurred.<sup>20</sup> The Jura Massif, situated on the foreland of the Alps, serves as a classic model for this process. It is a "fold-and-thrust belt" that formed as the sedimentary cover of Mesozoic limestones detached from its basement along a weak layer of Triassic evaporites and was pushed, folded, and stacked up by the advancing Alpine front.<sup>25</sup> It is this structural deformation that creates the dramatic topographic relief—the parallel ridges, deep valleys, and steep escarpments—that characterizes a limestone massif.<sup>20</sup> The specific style of this tectonic deformation is of profound importance, as it establishes the fundamental blueprint for the future evolution of the landscape. The large-scale structural weaknesses created during orogeny—the crests of anticlines, the troughs of synclines, and the extensive networks of faults and joints—become the primary pathways for water infiltration.<sup>8</sup> The subsequent process of chemical dissolution, or karstification, preferentially exploits these pre-existing lines of weakness.<sup>7</sup> Therefore, a folded range like the Jura develops a distinct karst drainage pattern, with underground rivers often aligned with the axes of synclines, which is markedly different from the patterns that might develop on a block-faulted plateau. In this way, the tectonic history of a massif dictates the large-scale architecture upon which all later erosional processes operate, controlling the evolution of its surface and subterranean landscapes.

# Part II: The Sculpting of Karst Landscapes

## **Section 3: The Chemistry of Dissolution**

#### 3.1 The Karstification Process

Once a limestone massif is uplifted and exposed to the atmosphere, it becomes subject to a unique form of chemical weathering known as karstification, named after the classic karst landscape of Slovenia.<sup>26</sup> This process is dominated by the dissolution of carbonate rock by slightly acidic water.<sup>7</sup> The primary agent is a weak carbonic acid (

H2CO3), which forms when atmospheric carbon dioxide (CO2) dissolves in rainwater. The chemical reactions governing this process are fundamental to understanding karst landscapes:

 Formation of Carbonic Acid: Rainwater absorbs CO2 from the atmosphere and becomes further enriched as it percolates through soil, where microbial and root respiration elevates CO2 concentrations.

2. Dissolution of Calcite: The weak carbonic acid reacts with the solid calcium carbonate (calcite) in the limestone, converting it into soluble calcium and bicarbonate ions, which are then carried away by the water.

While carbonic acid is the dominant force in most karst systems (epigenic karst), dissolution can also be driven by other acids. In some deep-seated systems (hypogenic karst), fluids rising from depth may contain sulfuric acid (H2SO4), formed by the oxidation of sulfide minerals, which is a much more aggressive solvent of carbonate rock.<sup>32</sup>

#### 3.2 Influential Factors

The intensity and character of karstification vary significantly depending on several interconnected factors:

• Climate and Water Availability: Abundant rainfall is the most critical requirement, as water is the medium for both the chemical reactions and the transport of dissolved

- rock.<sup>7</sup> Moderate to heavy rainfall promotes robust karst development.<sup>27</sup>
- **Bedrock Characteristics:** The purity, thickness, and structure of the limestone are paramount. Purer limestones with a higher percentage of calcite are more soluble. Dense, well-jointed, and thickly bedded rock provides an extensive network of fractures and bedding planes that act as initial conduits for water, greatly accelerating the dissolution process.
- Soil and Biogeochemistry: The role of the soil layer is crucial. Soils rich in organic
  matter host a vibrant community of microbes and plant roots that, through respiration
  and decomposition, can increase the CO2 concentration in soil air to levels many times
  that of the atmosphere. This significantly increases the acidity and dissolving power of
  infiltrating water.<sup>7</sup>
- **Topography and Hydraulic Gradient:** A significant difference in elevation across the massif (high topographic relief) creates a strong hydraulic head, which drives groundwater flow more vigorously through the rock mass, enhancing both chemical dissolution and mechanical erosion.<sup>7</sup>

The process of karstification thus represents a remarkable biogeochemical feedback system. The limestone rock itself is primarily a product of ancient life.<sup>14</sup> This ancient biogenic rock is then dissolved by carbonic acid, the formation of which is powerfully accelerated by the metabolic activities of modern life in the soil layer.<sup>7</sup> In this cycle, the biological carbon cycle actively drives the geological carbon cycle, creating a system where the products of past life are sculpted and broken down by the chemical byproducts of present life.

# **Section 4: Surface Manifestations (Exokarst)**

The slow, relentless process of dissolution carves a distinctive suite of landforms on the surface of limestone massifs, collectively known as exokarst. These features exist across a vast range of scales, from millimeters to kilometers.

#### 4.1 Microscale Features: The Formation of Karren and Lapiés

The initial interaction of acidic rainwater with exposed limestone creates an intricate micro-topography of solutional sculptures. <sup>26</sup> Known by the German term *Karren* or the French *lapiés*, these features include a variety of small-scale grooves, flutes, ridges, and pits etched into the rock surface. <sup>34</sup> Their morphology is dictated by the way water flows over the rock, with linear grooves (

*Rillenkarren*) forming on steeper slopes and shallow pans (*Kamenitzas*) developing on flatter surfaces where water can pool.<sup>26</sup> The appearance of these features can also indicate their formation environment; sharp, angular karren typically form on bare rock, while more rounded forms (

Rundkarren) are thought to develop beneath a soil cover before being exposed by erosion.<sup>35</sup>

These features collectively create a highly corrugated and rugged surface, often referred to as a limestone pavement, which consists of isolated blocks (clints) separated by widened fissures (grykes).<sup>35</sup>

#### 4.2 Mesoscale Features: The Development of Sinkholes (Dolines) and Uvalas

As dissolution continues, it becomes concentrated along larger structural weaknesses like joints and fractures, leading to the formation of one of the most characteristic karst landforms: the sinkhole, or doline.<sup>26</sup> These are enclosed depressions that can range in size from a few meters to over a kilometer in diameter.<sup>26</sup> They form through several distinct mechanisms:

- **Solution Dolines:** These form gradually as surface water funnels into a fissure, slowly dissolving the surrounding rock and creating a bowl-shaped depression.<sup>39</sup>
- **Collapse Dolines:** These form suddenly and often catastrophically when the roof of an underlying cave or void collapses, creating a steep-sided, often cylindrical depression.<sup>37</sup>
- Subsidence or Suffosion Dolines: These occur where a thick cover of soil or other unconsolidated sediment overlies the limestone. Infiltrating water washes this material down into underlying bedrock fissures, causing the surface to gradually slump or subside into a shallow depression.<sup>39</sup>

Over time, as individual dolines expand through continued dissolution and collapse, they may merge to form larger, compound, and irregularly shaped depressions known as uvalas.<sup>41</sup>

#### 4.3 Macroscale Landscapes: The Coalescence into Poljes

The largest of all karst depressions are poljes (a Serbo-Croatian word for "field"), which are extensive, flat-floored basins with steep, enclosing walls that can cover areas of up to 250 square kilometers. Their formation represents the most advanced stage of surface karst development, resulting from the large-scale coalescence of numerous sinkholes and uvalas, often guided by major tectonic structures such as faults or the axes of synclines. The floor of a polje is typically covered by a thick layer of soil derived from the insoluble residues of the dissolved limestone, making these features fertile and often the only areas suitable for agriculture in an otherwise barren, rocky landscape. A key feature of poljes is their unique hydrology; surface streams often flow across the flat floor before vanishing underground into sinkholes known as

*ponors* or "disappearing streams," and the entire basin may become a temporary lake during periods of high rainfall when the underground drainage system cannot accommodate the inflow.<sup>27</sup>

The progression from microscale karren to mesoscale dolines and finally to macroscale poljes illustrates a clear landscape evolution sequence. It demonstrates how a single fundamental process—dissolution—operating over different spatial and temporal scales, can generate a

distinct hierarchy of landforms. Dissolution begins by etching small channels into the rock surface; it then exploits and enlarges these pathways to create focused points of infiltration that develop into dolines; finally, over geological time, the amalgamation of these features along structural lines of weakness leads to the creation of vast, landscape-dominating poljes. This sequence reveals a clear evolutionary pathway where small, initial forms grow and merge into progressively larger and more complex landforms.

### Section 5: The Subterranean Realm (Endokarst)

While exokarst features shape the surface, the most dramatic transformations in a limestone massif occur underground, creating a complex, three-dimensional world of conduits and voids known as endokarst.

#### 5.1 Speleogenesis: The Formation of Caves and Underground River Networks

The process of cave formation, or speleogenesis, begins as acidic groundwater penetrates the rock mass through initial weaknesses like fractures, joints, and bedding planes. Over millennia, this water slowly dissolves the surrounding rock, gradually enlarging these small fissures into an interconnected network of passages, chambers, and vertical shafts. The development of this subterranean plumbing system is strongly influenced by the water table. The initial formation of major conduits often takes place in the phreatic zone—the region below the water table where all voids are completely filled with water. Here, water moves slowly under pressure, dissolving the rock in all directions to create characteristic rounded or elliptical passages. As regional river valleys deepen over geological time, the water table within the massif lowers, leaving these phreatic passages stranded in the vadose zone—the area above the water table. In this zone, water flows freely under the influence of gravity, carving stream channels into the cave floors and continuing to modify the cave's shape. In many cases, entire surface rivers are "captured" by the karst system, plunging into sinkholes or "swallow holes" to become powerful underground rivers that mechanically erode and further enlarge the cave system.

#### 5.2 The Science of Speleothems: A Detailed Typology of Cave Formations

Once cave passages are drained of water and filled with air, a second, depositional phase of development begins, leading to the formation of a spectacular array of secondary mineral deposits known as speleothems, or cave formations. <sup>44</sup> These are created by the reverse of the dissolution process. As water rich in dissolved calcium bicarbonate (Ca(HCO3)2) seeps into an air-filled cave, the partial pressure of CO2 in the cave atmosphere is typically lower than in the soil from which the water came. This causes CO2 to degas from

the water, making the calcium carbonate less soluble and causing it to precipitate out of solution, typically as calcite.<sup>13</sup>

The final morphology of a speleothem is a direct physical record of the specific hydrodynamic and geochemical conditions under which it formed. This principle allows for a detailed typology based on the movement of water:

- Dripstones: Formed by water dripping from the ceiling. A slow, steady drip creates a hollow, fragile tube called a "soda straw." If the tube becomes blocked, water flows over the outside, depositing layers of calcite to form the classic icicle-shaped stalactite. Water splashing on the floor below builds up a more massive, rounded stalagmite.
   When a stalactite and stalagmite meet, they form a column.<sup>13</sup>
- Flowstones: Formed by thin films of water flowing over surfaces. Water trickling down a sloped wall or ceiling creates curtain-like **shawls** or **draperies**. Broader sheets of water flowing across the floor or walls build up layers of **flowstone**, which can resemble frozen waterfalls. Slow-moving water on a slope can build low calcite walls known as **rimstone dams**, which impound pools of water.<sup>13</sup>
- Seepstones: Formed by water seeping very slowly through porous rock. Helictites are intricate, twisted formations that seem to defy gravity, growing in various directions as water moves through tiny internal channels via capillary action.<sup>44</sup> Nodular growths known as
  - **cave coral** or coralloids form from thin films of water moving over irregular surfaces, often influenced by air currents.<sup>46</sup>
- Pool Deposits: Formed in still bodies of water. Cave pearls are spherical concretions
  that form as calcite precipitates in concentric layers around a nucleus (like a grain of
  sand) that is kept in motion by dripping water. Thin, fragile sheets of calcite known as
  calcite rafts can form on the surface of calm pools.<sup>44</sup>

Beyond their beauty, speleothems are invaluable scientific archives. They can be precisely dated using radiometric techniques, and their growth layers, much like tree rings, contain isotopic and trace element information that can be used to reconstruct detailed records of past climate conditions, including temperature and rainfall.<sup>45</sup>

# Part III: A Global Tour of Prominent Limestone Massifs

The interplay between tectonic history, climate, and lithology produces a remarkable diversity of limestone massifs across the globe. Examining specific examples reveals how these fundamental processes result in unique geomorphological expressions and cultural landscapes.

# **Section 6: The European Karst Archetypes**

#### 6.1 Case Study: The Jura Mountains (France/Switzerland)

The Jura Mountains serve as the archetypal folded limestone massif, a classic thin-skinned fold-and-thrust belt formed during the Miocene as a direct result of the Alpine orogeny. The range is composed of thick sequences of Jurassic and Cretaceous limestones and marls that were detached from their crystalline basement and compressed into a series of parallel anticlines and synclines. This strong structural control has profoundly influenced its karst development. The landscape is characterized by numerous landforms that directly reflect the underlying geology, including anticlinal ridges, synclinal valleys, and dramatic steephead valleys known as

reculées, which are carved by resurgent rivers emerging from the base of limestone cliffs.<sup>25</sup> The region's high precipitation and fractured limestone have given rise to a mature and complex karst system, with over 10,000 documented caves, countless sinkholes, vast karren fields, and powerful karstic springs (resurgences) like the source of the Orbe River.<sup>50</sup> The hydrogeology is exceptionally complex, with rapid water transit through conduit networks, low natural filtration capacity, and a high vulnerability to surface pollution, making the management of its aquifers a critical issue.<sup>57</sup>

#### 6.2 Case Study: The Dolomites (Italy)

Recognized as a UNESCO World Heritage site, the Dolomites in the northern Italian Alps are of international significance for their spectacular geomorphology, representing the classic site for mountain development in dolomitic limestone.<sup>64</sup> Their landscape is defined by dramatic verticality, featuring sheer cliffs, towers, and pinnacles rising abruptly from gentle foothills.<sup>64</sup> The geological value of the Dolomites is exceptional, as they contain one of the world's best-preserved examples of Mesozoic carbonate platform systems.<sup>64</sup> These structures, often described as "fossilized atolls," were built by marine organisms in the Triassic period and provide a unique window into the recovery of marine life following the Permian-Triassic extinction event, the largest in Earth's history.<sup>64</sup> The current topography is a product of Alpine uplift followed by intense shaping by erosion, tectonism, and glaciation, which have sculpted the distinctive and monumental forms that define the "dolomitic landscape".<sup>64</sup>

#### **Section 7: The Diverse Karsts of the Americas**

7.1 Case Study: The Sierra Madre Oriental (Mexico)

The Sierra Madre Oriental is a major mountain range composed of a thick sequence of folded and faulted Mesozoic-age calcareous rocks, primarily limestones and dolomites.<sup>68</sup> This extensive range hosts a wide diversity of karst landscapes, from arid zones in the north to humid tropical environments in the south. The region is renowned for its spectacular endokarst features, including some of the deepest and largest caves in the Western Hemisphere.<sup>71</sup> A prime example is the

Sótano de las Golondrinas (Cave of Swallows), a colossal open-air pit cave with a free-fall drop of over 333 meters (1,092 feet).<sup>74</sup> Another significant feature is the Grutas de García, an extensive and ancient cave system (estimated at 50-60 million years old) made accessible by a cable car, showcasing large chambers and abundant speleothems.<sup>76</sup> The Sierra Gorda, a sub-region within the range, is part of the Huasteca Karst and is characterized by extremely rugged terrain with numerous deep pit caves known locally as sótanos, further highlighting the region's profound karst development.<sup>81</sup>

#### Section 8: The Vast and Varied Asian Massifs

#### 8.1 Case Study: South China Karst (Guilin) & Trang An (Vietnam)

The South China Karst, a UNESCO World Heritage site, is recognized as the world's type area for karst landform development in humid tropical and subtropical climates. The Guilin region, in particular, is the premier global example of mature tower karst (fenglin) and cone karst (fengcong). This iconic landscape of steep, isolated limestone towers rising from a flat plain represents the end stage of karst evolution. The "Guilin model" explains the coexistence of these forms as a function of the region's geological setting in a low-altitude basin that receives abundant allogenic water from surrounding areas, which promotes intense lateral erosion at the base of the towers. The nearby Trang An Landscape Complex in Vietnam is a related and equally spectacular landscape, featuring a maze of limestone karst peaks permeated by valleys, many of which are submerged by water, creating a landscape often called "Ha Long Bay on land".

#### 8.2 Case Study: The Limestone Massif (Syria)

The Limestone Massif in northwestern Syria, also known as the Belus Massif, is a highland region that stands as a profound example of a cultural karst landscape. It is famed for hosting the "Ancient Villages of Northern Syria," or "Dead Cities," a group of approximately 40 abandoned settlements dating from the 1st to 7th centuries. These villages, abandoned between the 8th and 10th centuries, provide a remarkably well-preserved testament to rural

life in late Antiquity and the Byzantine period.<sup>89</sup> The architectural remains of dwellings, pagan temples, churches, cisterns, and bathhouses illustrate the transition from the Roman pagan world to Byzantine Christianity. The relict landscape, with its evidence of hydraulic techniques and ancient agricultural plot plans, demonstrates the inhabitants' mastery of living and farming in a challenging karst environment.<sup>89</sup>

# Section 9: Unique Expressions in the Southern Hemisphere

#### 9.1 Case Study: The Nullarbor Plain (Australia)

The Nullarbor Plain is an immense, arid to semi-arid limestone plateau and the world's largest single exposure of limestone bedrock, covering about 200,000 square kilometers. <sup>90</sup> It represents a unique form of arid karst. Despite its vast size, its surface is remarkably flat and subdued, a result of the plain's low gradient and the high primary porosity of the limestone, which favored uniform downwasting rather than the development of concentrated drainage. <sup>90</sup> Its most distinctive karst features are subterranean or discrete. The plain is punctured by tens of thousands of "blowholes"—narrow vertical shafts that breathe air in response to barometric pressure changes—and several hundred caves, many accessed via large collapse dolines. <sup>90</sup> The cave systems, such as the extensive Old Homestead Cave, are often shallow and contain lakes of brackish or saline water. In the current arid climate, evaporite minerals like halite and gypsum are actively forming speleothems and contributing to rock breakdown through salt wedging, a process characteristic of desert karst environments. <sup>93</sup>

#### 9.2 Case Study: Tsingy de Bemaraha (Madagascar)

The Tsingy de Bemaraha National Park in Madagascar showcases one of the most spectacular and extreme karst landscapes on Earth. <sup>94</sup> The Malagasy word tsingy, meaning "where one cannot walk barefoot," aptly describes this veritable "forest" of razor-sharp limestone needles and pinnacles that can reach heights of over 100 meters. <sup>94</sup> This dramatic topography is the result of millions of years of intense erosion on a massive limestone plateau. Heavy rainfall has deeply dissected the rock along a dense network of vertical and horizontal fractures, dissolving the limestone to create a labyrinth of sharp ridges, deep canyons, and extensive underground cave systems. <sup>94</sup> This highly rugged and inaccessible terrain has created a natural fortress, protecting pockets of forest and providing a multitude of isolated microhabitats. As a result, the Tsingy is a sanctuary for an exceptionally large number of endemic and threatened species of plants and animals, making it a globally important center for biodiversity. <sup>95</sup>

Massif Name	Location	Dominant Karst	Tectonic Origin	Key Landforms	Notable
		Туре			Features
Jura	France /		Alpine Orogeny	Reculées,	Archetype of
Mountains	Switzerland		(Fold-and-Thru		structural
			st Belt)	synclines,	control on
				extensive	karst
				caves, karren	development;
				fields	complex
					hydrogeology. <sup>2</sup>
					5
The Dolomites	Italy	Alpine	Alpine Orogeny	Vertical walls,	UNESCO site;
		Dolomitic Karst		pinnacles,	"fossilized
			Deformation)	towers, spires,	atolls"
				glacial	preserving
				landforms	Triassic marine
					ecosystems.64
Sierra Madre	Mexico	Folded	Laramide	Deep canyons,	Contains some
Oriental		Mountain Karst	Orogeny	pit caves	of the world's
			(Fold-and-Thru	(sótanos),	deepest pit
			st Belt)	extensive cave	caves, like
				systems	Sótano de las
					Golondrinas. <sup>69</sup>
South China	China /	Humid Tropical	'	Tower karst	UNESCO site;
Karst	Vietnam	Tower Karst	of Carbonate	(fenglin), cone	global
		(Fenglin)	Platforms	karst	type-site for
				(fengcong),	humid tropical
				submerged	karst evolution
				valleys	("Guilin
					model"). <sup>83</sup>
	Syria	Plateau Karst	Regional Uplift	_	Famed for the
Massif				plateaus	"Dead Cities,"
					a UNESCO
					cultural
					landscape of
					ancient
					villages. <sup>1</sup>
	Australia	Arid Plateau	Miocene	Flat plain,	World's largest
Plain		Karst	Crustal Uplift	collapse	single
				dolines,	limestone
				blowholes,	exposure;
				shallow caves,	unique arid

				salt	karst
				speleothems	processes. <sup>90</sup>
Tsingy de	Madagascar	Tropical	Uplift of	"Forest" of	UNESCO site;
Bemaraha		Pinnacle Karst	Sedimentary	sharp	extreme
		(Tsingy)	Plateau	limestone	erosion forms
				needles, deep	a fortress-like
				canyons, caves	refuge for
					endemic
					biodiversity. <sup>94</sup>

# Part IV: Life in a Carbonate World: Karst Ecosystems and Biodiversity

Limestone massifs are not merely geological curiosities; they are vibrant, complex ecosystems. The unique chemical and physical properties of these landscapes create extreme environmental pressures that drive the evolution of highly specialized and often endemic life forms, both on the surface and deep within the subterranean realm.

# **Section 10: Specialized Ecosystems and Endemism**

Karst landscapes are globally recognized as biodiversity "hot spots". Their defining characteristic is extreme heterogeneity. A single massif can contain a mosaic of microhabitats, ranging from sun-baked, arid limestone pavements and cliffs to cool, moist, and shaded sinkhole bottoms, and finally to the perpetually dark, stable, and nutrient-poor environment of caves. This sharp environmental gradient over short distances, combined with the often-fragmented and isolated nature of karst outcrops, creates ideal conditions for evolutionary divergence and speciation. As a result, karst regions often harbor an extraordinary number of endemic species—organisms found nowhere else on Earth—and are considered "natural laboratories" for studying ecological and evolutionary processes.

#### **Section 11: The Calcicole Flora**

The soils that develop on limestone are chemically distinct, characterized by high pH (alkalinity), high concentrations of calcium ions, and often a deficiency in essential nutrients like phosphorus and iron, which become less soluble in alkaline conditions. Plants that are specially adapted to thrive in these conditions are known as **calcicoles**, or "lime-lovers". These species have evolved a suite of sophisticated

physiological adaptations to cope with their challenging environment:

- Managing Calcium Toxicity: To avoid the toxic effects of excess calcium, calcicoles employ several strategies. They can limit calcium uptake at the root level, actively sequester excess calcium ions within the large central vacuoles of their cells to isolate them from metabolic processes, and precipitate calcium as insoluble crystals of calcium oxalate.<sup>102</sup> This internal sequestration also helps with osmotic regulation in the often drought-prone, porous soils.<sup>102</sup>
- Acquiring Scarce Nutrients: To overcome the low availability of iron and phosphorus, calcicole roots exude organic compounds, such as carboxylates (citric and oxalic acids), that chelate iron and mobilize phosphorus, making them available for uptake.<sup>102</sup> Many calcicoles also form symbiotic relationships with mycorrhizal fungi, whose extensive hyphal networks are highly efficient at extracting these nutrients from the soil and transferring them to the plant.<sup>102</sup>

Examples of calcicole flora are found in families like Asteraceae (daisies) and Caryophyllaceae (pinks), as well as many rock-dwelling ferns of the genus *Asplenium*. These plants are characteristic components of unique habitats like European chalk grasslands and Mediterranean garrigue. Mediterranean garrigue.

# **Section 12: The Troglofauna**

The subterranean world of limestone caves represents one of the most extreme environments on Earth, defined by perpetual darkness, near-constant temperature and humidity, and an extreme scarcity of food resources.<sup>106</sup> Animals that have adapted to live exclusively in this environment are known as

**troglofauna** or troglobites.<sup>106</sup> Having evolved in complete isolation for millions of years, they exhibit a remarkable set of convergent evolutionary adaptations, known as troglomorphies:

- **Sensory Modifications:** In the absence of light, eyes and pigmentation are metabolically expensive and useless, so they are typically reduced or lost entirely. To compensate, other senses are heightened. Troglofauna often possess elongated appendages and antennae, which are covered in chemical and tactile receptors, allowing them to navigate, find food, and detect predators through touch and smell. To sense the sense of light, eyes and pigmentation are metabolically reduced or lost entirely.
- Metabolic Adaptations: The cave food web is based almost entirely on organic matter imported from the surface, primarily in the form of bat guano, or material washed in by streams.<sup>106</sup> This energy scarcity has driven the evolution of extremely low metabolic rates, which in turn lead to slower growth, delayed reproduction, and greatly increased lifespans compared to their surface-dwelling relatives.<sup>106</sup>

Troglofauna includes a wide array of invertebrates such as insects, spiders, pseudoscorpions, millipedes, and crustaceans, as well as some vertebrates like cave-dwelling salamanders and fish. Because cave systems are physically isolated from one another, troglofaunal populations cannot interbreed, leading to extremely high rates of endemism, with many

species being restricted to a single cave or a small group of interconnected caves. <sup>106</sup> The physical and chemical nature of limestone massifs creates powerful and distinct evolutionary pressures that result in two parallel forms of biological specialization. On the surface, the unique soil chemistry creates "edaphic islands," acting as a selective filter that permits only specially adapted calcicole flora to thrive, driving genetic divergence. <sup>102</sup> Simultaneously, beneath the surface, the dissolution of the rock creates physically isolated cave environments. This geographic isolation prevents gene flow and forces life to adapt to a completely different set of pressures—perpetual darkness and energy scarcity—leading to the evolution of highly specialized troglofauna. <sup>106</sup> In this way, the same geological substrate simultaneously drives two different modes of evolutionary isolation and specialization, one chemical and one physical, demonstrating the profound influence of geology on the patterns of life.

# Part V: Human-Karst Interactions: Resources, Risks, and Recreation

Limestone massifs are not just natural landscapes; they are also landscapes of profound human utility and significance. From providing the raw materials for civilization to supplying essential water resources and spaces for cultural and recreational activities, the relationship between humanity and these carbonate worlds is deep and complex.

# Section 13: Economic Foundations: Quarrying and Its Impacts

#### 13.1 The Limestone Industry

Limestone is arguably the world's most versatile and essential industrial mineral.<sup>3</sup> Its primary economic importance lies in the construction sector. When crushed, it is the most widely used rock aggregate for road beds, concrete, and asphalt.<sup>3</sup> It is also the indispensable raw material for the manufacture of Portland cement; limestone is heated in a kiln with clay to produce clinker, which is then ground into cement.<sup>3</sup> In agriculture, pulverized limestone (agricultural lime) is used extensively to neutralize acidic soils, improving fertility and crop yields.<sup>16</sup> Beyond these bulk uses, limestone's chemical properties make it vital for numerous industrial processes, including as a flux to remove impurities in steelmaking, in glass manufacturing, and as a filler in products as diverse as paper, paint, plastics, and even pharmaceuticals and food supplements.<sup>3</sup>

#### 13.2 Socio-Economic and Environmental Consequences

Limestone quarrying is a major economic driver, particularly in developing nations and rural areas where it provides direct employment and stimulates local economies through ancillary services. 118 Case studies from countries like Jamaica highlight the sector's contribution to GDP and its role in community development through the creation of jobs and infrastructure. 122 However, these benefits are often accompanied by significant negative impacts. Environmentally, quarrying involves the complete removal of vegetation and soil, habitat destruction, and fundamental alteration of the landscape. 119 Operations generate considerable dust and noise pollution, and blasting can cause vibrations that affect nearby communities. 3 Socially, while quarrying provides jobs, these can be hazardous, with risks of accidents and long-term health issues like respiratory diseases from dust exposure. 119 Furthermore, the economic benefits may not be evenly distributed, and local communities can become overly dependent on a finite resource, facing economic instability when a quarry closes. 118

# Section 14: The Hidden Resource: Karst Aquifers

#### 14.1 Global Importance

One of the most critical resources provided by limestone massifs is water. The interconnected network of conduits, fractures, and caves within karst formations creates highly productive aquifers that store and transmit vast quantities of groundwater. These aquifers are a primary source of freshwater for an estimated 25% of the world's population, supplying drinking water for major cities such as Vienna, Rome, Damascus, and San Antonio, and supporting agriculture and industry globally. Karst aquifers are characterized by rapid recharge, as surface water can directly enter the system through sinkholes and sinking streams, and by high discharge rates from large springs, which are often the source of major rivers.

#### 14.2 Management Challenges

The very features that make karst aquifers so productive also render them exceptionally vulnerable. The rapid, direct connection between the surface and the subsurface means there is little to no natural filtration of infiltrating water by soil and rock matrix. Oconsequently, pollutants such as agricultural fertilizers and pesticides, industrial chemicals, or urban runoff can be transported quickly and over long distances through the conduit network,

contaminating wells and springs with little warning.<sup>129</sup> The complex and often unpredictable nature of underground flow paths makes delineating catchment areas and establishing effective protection zones extremely challenging.<sup>129</sup> The Edwards Aquifer in Texas, USA, which is the sole source of drinking water for over two million people, serves as a prominent case study in the management of a critical karst resource. Its governance involves a complex framework of regulations, including the Edwards Aquifer Protection Program, which mandates protection plans for any construction or regulated activity over the aquifer's recharge and contributing zones to mitigate the risk of contamination.<sup>134</sup>

# **Section 15: Cultural and Recreational Landscapes**

#### 15.1 Limestone in Architecture

Throughout history, limestone has been a cornerstone of human architecture, valued for its durability, workability, and aesthetic appeal. Its use spans millennia and civilizations, from the monumental blocks of the Great Pyramids of Giza and the Great Sphinx to the travertine of the Roman Colosseum and the intricately carved facades of European Gothic cathedrals. In the modern era, it continues to be a prestigious material, cladding iconic structures like the Empire State Building and the Lincoln Memorial, lending them a sense of permanence and grandeur. The stone's use reflects not only its physical properties but also its cultural significance as a material that embodies strength, tradition, and civic importance.

#### 15.2 Tourism and Recreation

The dramatic and unique landscapes of limestone massifs make them popular destinations for tourism and recreation. The distinctive scenery of cliffs, gorges, and caves attracts millions of visitors annually for activities like hiking, rock climbing, and photography. The subterranean world offers a unique appeal for caving (also known as speleology or potholing), which ranges from guided tours in commercially developed "show caves" to technical exploration of wild cave systems. Tourism provides significant economic benefits to these often-rural regions. However, it also presents environmental challenges. High visitor numbers can lead to footpath erosion, soil compaction, and disturbance of wildlife. In sensitive cave environments, even the presence of humans can have negative impacts, such as the introduction of lint, microbes, and artificial light that can damage delicate speleothems and disrupt fragile ecosystems. The Yorkshire Dales National Park in the UK is a classic case study where management authorities must balance the demands of a high volume of tourism with the conservation of its renowned limestone landscape.

The human relationship with limestone massifs is defined by a central paradox: the very properties that make these landscapes so valuable are precisely what make them so fragile. The high permeability and conduit flow that create productive aquifers also make them conduits for pollution. The chemical purity that makes limestone a vital industrial resource is exploited through quarrying, a process that destroys the landscape and its hydrogeological function. The beautiful and mysterious caves that attract tourists are delicate ecosystems that can be irrevocably damaged by human presence. This inherent conflict means that any interaction with a limestone massif—whether for water, rock, or recreation—directly impacts its interconnected systems. Unlike more resilient landscapes, sustainable management is not merely an option but an absolute necessity to prevent irreversible degradation.

# Part VI: Conservation and Sustainable Management

The unique characteristics of limestone massifs present a formidable set of conservation challenges. Their fragility and the interconnectedness of their surface and subsurface environments demand specialized and holistic management strategies to mitigate threats and ensure their long-term sustainability.

#### **Section 16: Threats to Karst Environments**

Karst environments are among the most vulnerable ecosystems in the world due to a combination of geological and hydrological characteristics. The primary threats stem from human activities that fail to account for the landscape's unique sensitivity:

- Groundwater Pollution: This is the most pervasive threat. The rapid transit of water
  through conduits with minimal filtration means that contaminants from agricultural
  runoff (fertilizers, pesticides), industrial discharge, and inadequate wastewater disposal
  can swiftly and widely pollute entire aquifer systems, endangering drinking water
  supplies and subterranean ecosystems.<sup>129</sup>
- Land-Use Change and Degradation: Activities such as deforestation, intensive agriculture, and urbanization alter the natural hydrological regime. Increased impervious surfaces (e.g., roads, parking lots) concentrate runoff, which can accelerate erosion and trigger sinkhole collapse. The removal of vegetation and soil cover can lead to "rocky desertification," where the land becomes less productive and the underlying karst system is choked with sediment. Quarrying represents the most extreme form of land-use change, resulting in the complete destruction of the karst landscape and its associated habitats. 129
- Water Resource Over-extraction: Excessive pumping of groundwater from karst aquifers can lower the water table, causing springs and wells to dry up, threatening dependent ecosystems, and increasing the risk of land subsidence and sinkhole

- collapse.<sup>11</sup>
- Impacts on Cave Ecosystems: Beyond water pollution, direct human impacts on caves from unregulated tourism or vandalism can cause irreparable damage. The introduction of artificial lighting promotes the growth of invasive algae (*lampenflora*), while body heat, lint, and exhaled carbon dioxide can alter the delicate microclimate, damaging speleothems and harming sensitive troglofauna. 146

# **Section 17: Strategies for Protection and Stewardship**

Effective management of limestone massifs requires a paradigm shift from traditional, two-dimensional land management to a three-dimensional, systems-based approach that recognizes the inextricable link between the surface and the subsurface.

#### 17.1 International Frameworks: The Role of UNESCO Global Geoparks

The UNESCO Global Geoparks program provides an international framework for the holistic management of areas with significant geological heritage. A Geopark is not simply a protected area; it is a designation for a unified geographical area that manages its heritage with an integrated concept of protection, education, and sustainable development. This model promotes sustainable tourism, empowers local communities, supports scientific research, and raises awareness of geohazards. Geoparks like Shilin in China use their spectacular karst formations not only as a tourist attraction but also as an educational tool to inform the public about geology, conservation, and the cultural heritage of the indigenous communities, thereby fostering a sense of stewardship.

#### 17.2 National and Local Management: Case Studies

Effective conservation is often implemented at the national and local levels through dedicated management plans.

- The Yorkshire Dales National Park (UK): The management of this iconic limestone landscape is guided by a comprehensive National Park Management Plan, developed by a partnership of statutory bodies, local authorities, and community representatives. The plan sets strategic objectives to balance the demands of tourism, farming, and local economic well-being with the primary purpose of conserving and enhancing the area's natural beauty, wildlife, and cultural heritage. It is a model of collaborative governance for a living, working landscape.
- The Edwards Aquifer (USA): The protection of this critical karst aquifer is mandated by a robust regulatory framework. The Edwards Aquifer Protection Program, administered by the Texas Commission on Environmental Quality (TCEQ), requires detailed protection

plans for any development on the aquifer's recharge and contributing zones.<sup>134</sup> These plans must outline specific best management practices to prevent stormwater contaminants from reaching the aquifer. This approach is highly targeted, science-based, and legally enforceable, reflecting the high stakes of protecting a sole-source water supply for millions of people.<sup>135</sup>

#### 17.3 Recommendations for Sustainable Development in Karst Regions

Based on the challenges and successful management examples, a set of guiding principles for sustainable development in limestone massifs can be established. These include:

- **Comprehensive Assessment:** Prior to any development, thorough hydrogeological investigations, including dye tracing studies, are essential to map subsurface flow paths and delineate vulnerable areas.<sup>130</sup>
- Integrated Land-Use Planning: Zoning regulations must be specifically tailored to the vulnerabilities of karst terrain, restricting high-impact activities in critical recharge areas and establishing buffer zones around sensitive features like sinkholes and springs.<sup>11</sup>
- Best Management Practices: Implementation of strict guidelines for agriculture (e.g., minimizing fertilizer and pesticide use), industry (e.g., secure chemical storage), and urban development (e.g., advanced stormwater management) is crucial to prevent pollution at its source.<sup>129</sup>
- Public Education and Engagement: Fostering community awareness of the fragility of the karst environment and the connection between surface actions and groundwater quality is fundamental to building a culture of stewardship and ensuring the long-term success of conservation efforts.<sup>131</sup>

# Conclusion

# **Synthesizing the Multifaceted Nature of Limestone Massifs**

Limestone massifs are among the most dynamic and complex landscapes on Earth. They are geological archives, born from the accumulated remains of ancient marine life and uplifted by the immense power of tectonic forces. They are sculpted by a unique and relentless process of chemical dissolution that creates a dual world of spectacular surface topography and a vast, hidden subterranean realm. This intricate geomorphology gives rise to highly specialized ecosystems, making these massifs critical reservoirs of global biodiversity. For humanity, they are landscapes of paradox: they provide indispensable resources such as building materials and fresh water, yet they are exceptionally fragile. The very properties of porosity and

solubility that make them valuable are the same properties that render them acutely vulnerable to pollution, degradation, and collapse. Their management, therefore, requires a profound understanding of the intimate and immediate connection between surface activities and the unseen subsurface environment.

# Future Outlook: Research Directions and Emerging Conservation Challenges

Looking forward, the sustainable stewardship of limestone massifs faces significant challenges, particularly in the context of global environmental change. Future research must focus on understanding the impacts of a changing climate on karst systems. Altered precipitation patterns and rising temperatures will affect aquifer recharge rates, dissolution kinetics, and the stability of sensitive cave microclimates and ecosystems. There is an urgent need for more sophisticated groundwater models that can accurately predict contaminant transport in complex conduit systems to better inform land-use planning and water resource protection. The conservation of these landscapes will increasingly depend on integrated management strategies, like those promoted by UNESCO Global Geoparks, that successfully balance ecological protection, sustainable economic development, and community well-being. Ultimately, ensuring the future of these remarkable carbonate worlds requires a shift in human perspective—from simply living on karst to learning how to live in harmony with it.

# **Works Cited**

- 1. <sup>1</sup> Wikipedia. (n.d.). *Limestone Massif*.
- 2. <sup>3</sup> U.S. Geological Survey. (2008). Limestone—A Crucial and Versatile Industrial Mineral Commodity.
- 3. <sup>4</sup> Arkansas Geological Survey. (n.d.). *Limestone*.
- 4. <sup>12</sup> Calanques National Park. (n.d.). *Geology and landscapes.*
- 5. <sup>13</sup> Domain of the Caves of Han. (n.d.). *Understanding the cave/geology.*
- 6. <sup>17</sup> World of Stones USA. (n.d.). What is Limestone & How it Formed.
- 7. <sup>14</sup> Kansas Geological Survey. (n.d.). *Limestone*.
- 8. <sup>16</sup> Baker Lime. (n.d.).

How Limestone Is Formed.

9. <sup>9</sup> EBSCO. (n.d.).

Limestone.

10. <sup>21</sup> Wikipedia. (n.d.).

Lithification.

11. <sup>22</sup> Reddit. (2012).

How does lithification happen in nature and how long does it take?.

12. 18 Britannica. (n.d.).

Limestone.

13. <sup>25</sup> Wikipedia. (n.d.).

Geology of the Jura Massif.

14. <sup>23</sup> Ollier, C. D., & Pain, C. F. (2009). Mountain uplift and the Neotectonic Period. *ResearchGate*.

15. <sup>24</sup> Wikipedia. (n.d.).

Massif Central.

16. <sup>7</sup> Earle, S. (n.d.).

Karst Landscapes and Systems. Thompson Rivers University.

17. 8 Earle, S. (n.d.).

Karst Landscapes and Systems. LibreTexts Geo.

18. 37 National Park Service. (n.d.).

Karst Landscapes.

19.  $^{38}$  Wisconsin Geological and Natural History Survey. (n.d.).

Karst and Sinkholes.

20. 40 British Geological Survey. (n.d.).

Sinkholes.

21. 26 INFLIBNET Centre. (n.d.).

The Karst Landforms and Cycle of Erosion.

22. 44 Australian Museum. (2018).

Limestone caves.

23. 29 U.S. Geological Survey. (n.d.).

Caves.

24. 30 British Geological Survey. (n.d.).

How caves form.

25. <sup>45</sup> National Centers for Environmental Information. (n.d.). *Speleothem*.

26.  $^{47}$  National Centers for Environmental Information. (n.d.).

Speleothem.

27. 46 Missouri State Parks. (n.d.).

Speleothems.

28. <sup>2</sup> Ruellan, A. (n.d.).

Les sols calcaires. IRD Documentation.

29. <sup>5</sup> Office québécois de la langue française. (1986).

Calcaire.

30. 6 Wikipedia. (n.d.).

Calcaire.

31. 20 Parc naturel régional de Chartreuse. (2017).

Une montagne calcaire.

32. 15 Odyssée de la Terre. (n.d.).

Reliefs karstiques: quand l'eau sculpte le calcaire.

33. 19 Assistance Scolaire Personnalisée. (n.d.).

La formation des roches sédimentaires.

34. <sup>28</sup> StudySmarter. (n.d.).

Proceso kárstico.

35. 158 Ingeoexpert. (2019).

¿Qué es el relieve kárstico y cómo se forma?.

36. 159 ResearchGate. (2016).

Procesos de karstificación.

37. 39 Bordino, J. (2025).

Dolinas o torcas: qué son, formación, tipos y ejemplos. GEOenciclopedia.

38. 160 Figueroa, G. (2021).

Dolina - Geología. JovaGeology.

39. 42 Fiveable. (n.d.).

Polie.

40. 27 Britannica. (n.d.).

polie.

41. 41 Homeschool.ie. (2020).

Leaving Certificate Geography: Karst Landscapes.

42. <sup>43</sup> Hughes, P. D., et al. (2013).

Pleistocene fluvial archives in the karst poljes of the Dinaric Alps. Geological Society of America Abstracts with Programs.

43. 35 Britannica. (n.d.).

lapiés.

44. <sup>34</sup> Fiveable. (n.d.).

Karren.

45. <sup>36</sup> Burrenbeo Trust. (n.d.).

Geology.

46. 161 Veress, M. (2014).

Karren of High Mountains. Hungarian Geographical Bulletin.

47. 32 Wikipedia. (n.d.).

Karst.

48. <sup>7</sup> Earle, S. (n.d.).

Karst Landscapes and Systems. Thompson Rivers University.

49. 31 City of Austin. (2011).

Karst Lesson Plan.

50. 33 Fiveable. (n.d.).

Karst Processes & Dissolution Chemistry.

51. 1 Wikipedia. (n.d.).

Limestone Massif.

52. 88 UNESCO World Heritage Centre. (n.d.).

Trang An Landscape Complex.

53. 89 UNESCO World Heritage Centre. (n.d.).

Ancient Villages of Northern Syria.

54. 162 Oxalis Adventure. (n.d.).

10 Most Famous Caves in the World.

55. 163 Odyssée de la Terre. (n.d.).

Les 10 plus belles grottes du monde.

56. 164 Tamera. (n.d.).

Immersion dans les régions karstiques du nord Vietnam.

57. 165 Fédération Française de Spéléologie. (n.d.).

Les grands massifs karstiques de France.

58. 166 Agence de l'eau Rhône-Méditerranée Corse. (2018).

Les formations calcaires et le karst.

59. 167 FD Minerals. (2018).

Caliza.

60. 168 Pulycort. (n.d.).

11 Tipos de piedra Caliza y Arenisca.

61. 169 Geo-Sports.org. (2024).

Étape 8: Histoire de la construction.

62. 170 Conversa Online. (n.d.).

Piedras Calizas y Mármoles en la Historia.

63. 64 UNESCO World Heritage Centre. (n.d.).

The Dolomites.

64. 67 Dolomiti Bellunesi National Park. (n.d.).

Geology.

65. 65 Bosellini, A., et al. (2003). The Triassic reefs and carbonate platforms of the Dolomites, northern Italy.

Episodes.

66. 66 South-Tirol.com. (n.d.).

Formation of the Dolomites.

67. <sup>49</sup> Madritsch, H., et al. (2010). Deciphering neotectonics from river profile analysis in the karst Jura Mountains.

ResearchGate.

68. <sup>57</sup> Pronk, M., et al. (2015). Twenty years of monitoring of karst groundwater quality in the

Swiss Jura Mountains.

PubMed.

69. <sup>58</sup> SOKARST. (n.d.).

Jurassic Karst.

70. <sup>48</sup> France.fr. (n.d.).

Jura Mountains.

71. <sup>68</sup> Wikipedia. (n.d.).

Sierra Madre Occidental.

72. 171 ResearchGate. (2013).

Compilation and Production of a Karst map of Mexico.

73. <sup>69</sup> Weidie, A. E., & Murray, G. E. (1967). Geology of the Canon De La Huasteca Area in the Sierra Madre Oriental, Nuevo Leon, Mexico.

Tulane Studies in Geology and Paleontology.

74. 83 UNESCO World Heritage Centre. (n.d.).

South China Karst.

75. 85 IUGS International Commission on Geoheritage. (n.d.).

Guilin Karst.

76. 86 Natural World Heritage Sites. (2020).

South China Karst.

77. 87 TravelChinaGuide.com. (n.d.).

South China Karst.

78. 84 Wikipedia. (n.d.).

South China Karst.

79. 172 Department of Conservation, New Zealand. (n.d.).

Limestone ecosystems.

80. <sup>173</sup> D'Angeli, I. M., et al. (2024). The geomicrobiology of limestone, sulfuric acid speleogenetic, and volcanic caves.

PubMed Central.

81. 174 UNESCO Man and the Biosphere Programme. (n.d.).

Sierra de Grazalema.

82. 175 Center for Nature Conservation and Development. (n.d.).

The Northern Limestone Conservation Landscape.

83. <sup>176</sup> Braholli, E. (2022). Karst and ecosystem service in the large limestones massif of Mali i Thatë-Rakickë.

ResearchGate.

84. <sup>98</sup> Eme, D., et al. (2023). Biodiversity in Karst Landscapes: Introduction to the Special Issue.

MDPI.

85. 177 NSW Government. (n.d.).

Karst and caves.

86. 99 Eme, D., et al. (2023). Biodiversity in Karst Landscapes: Introduction to the Special

Issue.

ResearchGate.

87. 101 Frontiers in Plant Science. (n.d.).

Karst Plants Diversity, Evolution, Taxonomy and Conservation.

88. 100 MDPI. (n.d.).

Special Issue: Biodiversity in Karst Landscapes.

89. 102 Wikipedia. (n.d.).

Calcicole.

90. <sup>178</sup> Salisbury, E. J. (1920). The Significance of the Calcicolous Habit. *Journal of Ecology*.

91. <sup>179</sup> White, P. J., & Broadley, M. R. (2021). The secrets of calcicole species revealed. *ResearchGate*.

92. <sup>103</sup> Tyler, G. (2002). Immobilization of tissue iron on calcareous soil: differences between calcicole and calcifuge plants.

Lund University.

93. <sup>180</sup> White, P. J., & Broadley, M. R. (2021). The secrets of calcicole species revealed. *Journal of Experimental Botany*.

94. 106 Wikipedia. (n.d.).

Troglofauna.

95. 112 Bennelongia Environmental Consultants. (n.d.).

Troglofauna (Subterranean Fauna).

96. 109 Wikipedia. (n.d.).

List of troglobites.

97. <sup>113</sup> Halse, S., et al. (2014). Troglofauna in the vadose zone: comparison of scraping and trapping results and sampling adequacy.

Subterranean Biology.

98. 114 YouTube. (2020).

Troglofauna: Animals of the Caves.

99. 181 Texas Parks & Wildlife Department. (n.d.).

Cave Creatures - Vocabulary.

100. 107 Natural History Museum. (n.d.).

Cave life: creatures that lurk in the shadows.

101. <sup>111</sup> Price, L. (2004). An Introduction to Some Cave Fauna of Malaysia and Thailand. *Digital Commons @ USF*.

102. <sup>182</sup> Oregon Grotto. (n.d.).

Fauna of Northwest caves.

103. 108 Mossy Earth. (n.d.).

Extraordinary and Endangered Cave Dwelling-Animals.

104. <sup>183</sup> Turismo de Grazalema. (n.d.).

Fauna.

105. <sup>184</sup> Wikipedia. (n.d.).

Macizo de Larra-Belagua.

106. <sup>185</sup> Grand Site de France Solutré Pouilly Vergisson. (n.d.).

Los prados de piedra caliza.

107. <sup>186</sup> Nhan Dan Online. (2024).

Contemplar reserva natural del humedal de Van Long.

108. <sup>187</sup> Ayuntamiento de Orihuela. (n.d.).

Sierra de Orihuela.

109. <sup>188</sup> DREAL PACA. (2012).

Éboulis calcaires et de schistes calcaires des étages montagnard à alpin.

110. <sup>189</sup> StudySmarter. (n.d.).

Massifs calcaires: Définition & Formation.

111. To Lesser, J. M., & Lesser, G. (1988). Region 9, Sierra Madre Oriental. In *Hydrogeology*. Geological Society of America.

112. <sup>68</sup> Wikipedia. (n.d.).

Sierra Madre Occidental.

113. <sup>190</sup> Mahlknecht, J., et al. (2018). Modeling of groundwater processes in a karstic aquifer of Sierra Madre Oriental, Mexico.

ResearchGate.

114. <sup>69</sup> Weidie, A. E., & Murray, G. E. (1967). Geology of the Canon De La Huasteca Area in the Sierra Madre Oriental, Nuevo Leon, Mexico.

Tulane Studies in Geology and Paleontology.

115. <sup>54</sup> ResearchGate. (2020).

Structural and Karstic Landscapes of the Joux Valley, Southwestern Jura.

116. <sup>50</sup> ResearchGate. (2020).

Structural and Karstic Landscapes of the Joux Valley, Southwestern Jura.

117. <sup>191</sup> Geopark Schwäbische Alb. (n.d.).

What is the Jura.

118. <sup>53</sup> Montagnes du Jura. (n.d.).

Top 7 most spectacular landscapes.

119. 117 Pontotoc Sand & Stone. (n.d.).

Limestone in Different Industries.

120. <sup>115</sup> British Geological Survey. (2006).

Mineral Planning Factsheet: Limestone.

121. <sup>3</sup> U.S. Geological Survey. (2008).

Limestone—A Crucial and Versatile Industrial Mineral Commodity.

122. <sup>116</sup> Derbyshire County Council. (2022).

Industrial Limestone Background Paper.

123. <sup>10</sup> WHYMAP. (n.d.).

World Karst Aquifer Map.

124. <sup>11</sup> Stevanović, Z. (Ed.). (2024).

Karst: Environment and Management of Aquifers. The Groundwater Project.

125. 127 The Watershed Association. (n.d.).

Karst.

126. 128 Karst Waters Institute. (n.d.).

Water Resources.

127. 143 Stoneco. (n.d.).

The Benefits of Using Limestone.

128. <sup>142</sup> Internet Geography. (n.d.).

Limestone and Recreation.

129. 144 Washington and Lee University Outing Club. (n.d.).

Caving.

130. 145 British Geological Survey. (n.d.).

How caves form.

131. <sup>130</sup> Government of British Columbia. (2024).

Karst Management.

132. <sup>150</sup> Maryland Geological Survey. (n.d.).

Engineering Problems in Karst.

133. <sup>129</sup> Fiveable. (n.d.).

Environmental Issues in Karst Regions.

134. <sup>132</sup> Minnesota Pollution Control Agency. (n.d.).

Protecting water in karst regions.

135. <sup>151</sup> Global Geoparks Network. (2023).

UNESCO Global Geoparks.

136. <sup>152</sup> Wikipedia. (n.d.).

UNESCO Global Geoparks.

137. 153 UNESCO. (n.d.).

Shilin UNESCO Global Geopark.

138. <sup>192</sup> Tien-Shan Geological Society. (n.d.).

Geoparks - the basis of integrated conservation of ecosystems.

139. <sup>102</sup> Wikipedia. (n.d.).

Calcicole.

140. 105 Wikipedia. (n.d.).

Calcicole.

141. <sup>104</sup> Tyler, G. (2002). Inorganic phosphorus fractions of litter and humus in relation to soil acidity and plant species composition in a South Swedish deciduous forest. *Annals of Botany*.

142. <sup>193</sup> Edinburgh Garden School. (2015).

Calcifuges and Calcicoles.

143. 109 Wikipedia. (n.d.).

List of troglobites.

144. <sup>110</sup> Giachino, P. M., & Vailati, D. (2010). Subterranean (Hypogean) Habitats in Karst and Their Fauna.

ResearchGate.

145. <sup>194</sup> Wikipedia. (n.d.).

Subterranean fauna.

146. 146 National Park Service. (n.d.).

Cave Impacts.

147. National Park Service. (n.d.).

Cave Life.

148.  $^{71}$  Bautista, F., et al. (2021). The karst landscapes of Mexico.

ResearchGate.

149. <sup>68</sup> Wikipedia. (n.d.).

Sierra Madre Occidental.

150. <sup>73</sup> Villaseñor, J. L., et al. (2023). Vascular plant richness in the mountain karst forests of Mexico.

bioRxiv.

151. 195 PeakVisor, (n.d.).

Sierra Madre Oriental.

152. <sup>50</sup> Reynard, E., & Schoeneich, P. (2020). Structural and Karstic Landscapes of the Joux Valley, Southwestern Jura.

ResearchGate.

153. <sup>25</sup> Wikipedia. (n.d.).

Geology of the Jura Massif.

154. <sup>48</sup> France.fr. (n.d.).

Jura Mountains.

155. <sup>196</sup> Réserve Naturelle Nationale de la Haute Chaîne du Jura. (n.d.). *Geology*.

156. <sup>197</sup> Boukhelkhal, A., & Ozer, P. (2020). A case study of the life cycle impact of limestone quarrying on the environment.

International Journal of Global Warming.

157. <sup>118</sup> Ogbonna, C. C., et al. (2015). Communities-Company Relation in Limestone Mining Region of Southwest Nigeria.

European-American Journals.

158. <sup>119</sup> Pramanik, S., & Das, B. (2023). Environmental and Socio-Economic Impacts of Stone Quarrying: A Case Study.

Humanities, Social Sciences and Statesman.

159. <sup>125</sup> Khan, W. D., et al. (2021). Effects of Limestone Quarry Solid Waste on the Growth of Two Native Plant Species.

Frontiers in Ecology and Evolution.

160. 133 Minnesota Pollution Control Agency. (n.d.).

Guidance for Investigation and Closure of Karst Features.

161. <sup>10</sup> WHYMAP. (n.d.).

World Karst Aquifer Map.

162. <sup>11</sup> Stevanović, Z. (Ed.). (2024).

Karst: Environment and Management of Aquifers. The Groundwater Project.

163. 198 U.S. Geological Survey. (2021).

Karst Aquifers.

164. <sup>199</sup> Fernandes, L. F. S., et al. (2020). The Impact of Tourism Activity on Coastal Biodiversity: A Case Study at Praia da Cova Redonda, Algarve-Portugal. *ResearchGate*.

165. <sup>148</sup> Internet Geography. (n.d.).

Limestone Case Study - Malham, The Yorkshire Dales.

166. <sup>200</sup> Fernandes, L. F. S., et al. (2020). The Impact of Tourism Activity on Coastal Biodiversity.

MDPI.

167. <sup>201</sup> EuroLithos. (n.d.).

Case Studies.

168. <sup>150</sup> Maryland Geological Survey. (n.d.).

Engineering Problems in Karst.

169. <sup>131</sup> Cave Conservancy of the Virginias. (n.d.).

Karst.

170. 177 NSW Government. (n.d.).

Karst and caves.

171. <sup>149</sup> Parise, M. (2011). Karst Environments: Problems, Management, Human Impacts, and Sustainability.

ResearchGate.

172. <sup>90</sup> Webb, J. A., & James, J. M. (2006). Karst evolution of the Nullarbor Plain, Australia.

ResearchGate.

173. 93 Western Australian Speleological Group. (n.d.).

Nullarbor Karst.

174. <sup>92</sup> Lipar, M., & Ferk, M. (2011). Origin and karst geomorphological significance of the Nullarbor Plain blowholes.

Earth Surface Processes and Landforms.

175. <sup>91</sup> Wikipedia. (n.d.).

Nullarbor Plain.

176. <sup>94</sup> Aqua-Firma. (n.d.).

Tsingy de Bemaraha, Madagascar.

177. <sup>96</sup> Atlas Obscura. (2024).

Madagascar.

178. 95 Wikipedia. (n.d.).

Tsingy de Bemaraha National Park.

179. <sup>202</sup> UNEP-WCMC. (n.d.).

South China Karst.

180. <sup>203</sup> One Earth. (n.d.).

Sierra Madre Oriental Pine-Oak Forests.

181. <sup>72</sup> Wikipedia. (n.d.).

Sierra Madre Oriental.

182. <sup>204</sup> Elovoyage. (n.d.).

Mexico: The 5 Most Beautiful Cenotes in the Yucatan.

183. <sup>81</sup> Wikipedia. (n.d.).

Sierra Gorda.

184. 82 UNESCO Man and the Biosphere Programme. (n.d.).

Sierra Gorda.

185. <sup>55</sup> Luetscher, M., & Jeannin, P. F. (2004). Ice caves as an indicator of winter climate evolution: A case study from the Jura Mountains. *ResearchGate*.

186. <sup>205</sup> Wikipedia. (n.d.).

Jura Mountains.

187. <sup>206</sup> Polish-Jura.com. (n.d.).

Speleology - caves' exploring.

188. <sup>56</sup> Montagnes du Jura. (n.d.).

Top 5 caves, caverns and chasms in the Jura Mountains.

189. <sup>207</sup> Britannica. (n.d.).

Swabian Alp.

190. <sup>208</sup> Montagnes du Jura. (n.d.).

Sentier karstique du grand bois et grotte Maëva.

191. <sup>59</sup> ZAAJ. (n.d.).

The Jurassic Karst Hydrogeological Observatory.

192. 60 Goldscheider, N., et al. (2015).

Overview of Methods Applied in Karst Hydrogeology.

193. 120 StudyRocket. (n.d.).

The social, economic and environmental benefits and drawbacks of limestone quarrying.

194. <sup>121</sup> Mbilinyi, B., et al. (2025). Livelihood dynamics of rural households in quarrying communities.

Cogent Social Sciences.

195. 139 Outside Stone. (n.d.).

Exploring the history of limestone in architecture and its usage over time.

196. <sup>140</sup> Architextures. (n.d.).

Limestone.

197. <sup>141</sup> Regatta Exports. (n.d.).

Famous Buildings That Are Made Using Limestone.

198. <sup>126</sup> WHYMAP. (n.d.).

World Karst Aquifer Map.

199. 209 Minnesota Protect Our Water. (n.d.).

Karst Features and Aquifers in Minnesota.

200. <sup>11</sup> Stevanović, Z. (Ed.). (2024).

Karst: Environment and Management of Aquifers. The Groundwater Project.

201. <sup>210</sup> Day, M. J., & Hall, C. M. (2014). Ecotourism in the State Forest Karst of Puerto Rico.

Journal of Cave and Karst Studies.

202. <sup>211</sup> Day, M. J., & Hall, C. M. (2014). Ecotourism in the State Forest Karst of Puerto Rico. *ResearchGate*.

203. <sup>212</sup> Sunardi, et al. (2021). A Framework for Developing Coastal Green Tourism in the Gunungsewu Geopark, Indonesia.

International Journal of Geomate.

204. <sup>64</sup> UNESCO World Heritage Centre. (n.d.).

The Dolomites.

205. <sup>213</sup> IUCN World Heritage Outlook. (n.d.).

The Dolomites.

206. <sup>214</sup> IUCN. (2017).

The Dolomites - 2017 Conservation Outlook Assessment.

207. <sup>215</sup> IUCN. (n.d.).

IUCN Red List Habitat Classification.

208. <sup>216</sup> IUCN. (2012).

Habitats Classification Scheme (Version 3.1) Guidance.

209. <sup>217</sup> IUCN. (n.d.).

Habitats Classification Scheme (Version 3.1).

210. <sup>218</sup> One Earth. (n.d.).

Nullarbor Plains Xeric Shrublands.

211. <sup>76</sup> Nuevo León Travel. (n.d.).

Grutas de García.

212. <sup>77</sup> Wikipedia. (n.d.).

Grutas de García.

213. <sup>78</sup> Wikipedia. (n.d.).

Grutas de García.

214. <sup>79</sup> Grutas de Garcia. (2023).

Grutas de Garcia.

215. 80 GetYourGuide. (n.d.).

Grutas de García.

216. Russell, W. H., & Terry, C. (1968). Sotano de las Golondrinas.

AMCS Bulletin.

217. <sup>75</sup> Wikipedia. (n.d.).

Cave of Swallows.

218. <sup>219</sup> Travesías Digital. (n.d.).

Sótano de las Golondrinas en la Huasteca Potosina.

219. <sup>220</sup> Fendley, S., et al. (2021). Development and implementation of virtual field teaching resources.

ResearchGate.

220. <sup>221</sup> RopeWiki. (n.d.).

Sotano de las Golondrinas.

221. <sup>51</sup> Agence de l'eau Rhône-Méditerranée Corse. (n.d.).

Calcaires du Jurassique moyen du premier plateau du Jura.

222. <sup>52</sup> CPEPESC. (2003).

Les karsts jurassiens.

223. <sup>61</sup> Epage Doubs Dessoubre. (n.d.).

Le karst et les rivières du massif jurassien.

224. <sup>62</sup> MagniJura. (n.d.).

Le Karst.

225. <sup>63</sup> ORISK-BFC. (2021).

Karst et karsts régionaux.

226. 122 Development Minerals Programme. (n.d.).

Jamaica.

227. <sup>222</sup> High Commission of India, Kingston. (n.d.).

Market Survey Report on Mining & Exportation of Limestone.

228. <sup>123</sup> Mines and Geology Division, Jamaica. (2024).

Jamaica Mining Sector Performance 2023.

229. <sup>223</sup> JAMPRO. (2016).

Terms of Reference: Feasibility Study on the Limestone Sector's Value Chain.

230. 124 Planning Institute of Jamaica. (2009).

Vision 2030 Jamaica: Mining and Quarrying Sector Plan.

231. <sup>134</sup> Texas Commission on Environmental Quality. (n.d.).

Edwards Aquifer Protection Plans.

232. <sup>135</sup> Greater Edwards Aquifer Alliance. (n.d.).

Edwards Aquifer Protection Plan.

233. <sup>136</sup> Texas Commission on Environmental Quality. (n.d.).

Edwards Aquifer Protection Program.

234. <sup>137</sup> Ecosystem Planning & Restoration. (n.d.).

Edwards Aquifer Monitoring and Research Plan.

235. <sup>138</sup> Edwards Aquifer Authority. (n.d.).

Managing The Edwards Aguifer.

236. <sup>154</sup> Yorkshire Dales National Park Authority. (n.d.).

Tourism in the Dales.

237. 155 Yorkshire Dales National Park Authority. (2024).

Draft SA Report.

238. <sup>224</sup> Yorkshire Dales National Park Authority. (n.d.).

The new Management Plan 2025-2030.

239. <sup>156</sup> Yorkshire Dales National Park Authority. (2021). Yorkshire Dales National Park Management Plan 2019-24.

240. <sup>157</sup> Yorkshire Dales National Park Authority. (2025). *Progress on National Park plans*.

241. <sup>218</sup> One Earth. (n.d.).

Nullarbor Plains Xeric Shrublands.

242. <sup>225</sup> Australian Government. (n.d.). *Nullarbor Bioregion*.

243. <sup>226</sup> Journey Beyond Rail. (n.d.). *The Nullarbor: Anything but plain.* 

244. <sup>227</sup> One Earth. (n.d.).

Nullarbor Plains Xeric Shrublands.

245. <sup>97</sup> African World Heritage Sites. (n.d.). *Tsingy de Bemaraha, Madagascar*.

246. <sup>44</sup> Australian Museum. (2018). *Limestone caves*.

247. <sup>37</sup> National Park Service. (n.d.). *Karst Landscapes*.

248. <sup>25</sup> Wikipedia. (2025).

Geology of the Jura Massif.

249. <sup>15</sup> Odyssée de la Terre. (2025). Reliefs karstiques: quand l'eau sculpte le calcaire.

250. <sup>39</sup> GEOenciclopedia. (n.d.).

Dolinas o torcas: qué son, formación, tipos y ejemplos.

251. <sup>44</sup> Australian Museum. (2018). *Limestone caves*.

252. <sup>27</sup> Britannica. (n.d.). polie.

253. <sup>35</sup> Britannica. (n.d.). *lapiés*.

254. <sup>64</sup> UNESCO World Heritage Centre. (n.d.). *The Dolomites*.

255. <sup>228</sup> France.fr. (n.d.).

Jura Mountains.

256. <sup>83</sup> UNESCO World Heritage Centre. (n.d.). *South China Karst*.

257. <sup>102</sup> Wikipedia. (2025).

Calcicole.

258. <sup>106</sup> Wikipedia. (2025). *Troglofauna*.

259. The Lesser, J. M., & Lesser, G. (1988). Region 9, Sierra Madre Oriental. In *Hydrogeology*. Geological Society of America.

260. <sup>50</sup> Reynard, E., & Schoeneich, P. (2020). Structural and Karstic Landscapes of the Joux Valley, Southwestern Jura.

ResearchGate.

261. 116 Derbyshire County Council. (2022).

Industrial Limestone Background Paper.

262. <sup>11</sup> Stevanović, Z. (Ed.). (2024).

Karst: Environment and Management of Aguifers. The Groundwater Project.

263. <sup>142</sup> Internet Geography. (n.d.).

Limestone and Recreation.

264. <sup>130</sup> Government of British Columbia. (2024).

Karst Management.

265. <sup>90</sup> Webb, J. A., & James, J. M. (2006). Karst evolution of the Nullarbor Plain, Australia.

ResearchGate.

266. <sup>94</sup> Aqua-Firma. (n.d.).

Tsingy de Bemaraha, Madagascar.

267. 67 Dolomiti Bellunesi National Park. (n.d.).

Geology.

268. <sup>202</sup> UNEP-WCMC. (n.d.).

South China Karst.

#### Works cited

- 1. Limestone Massif Wikipedia, accessed August 31, 2025, https://en.wikipedia.org/wiki/Limestone Massif
- 2. Les sols calcaires : les principaux travaux des pédologues français Horizon IRD, accessed August 31, 2025,

https://horizon.documentation.ird.fr/exl-doc/pleins\_textes/divers15-02/21624.pdf

- 3. Limestone, as used by the minerals USGS Publications Warehouse, accessed August 31, 2025, <a href="https://pubs.usgs.gov/fs/2008/3089/fs2008-3089.pdf">https://pubs.usgs.gov/fs/2008/3089/fs2008-3089.pdf</a>
- 4. Limestone Arkansas Geological Survey, accessed August 31, 2025, <a href="https://www.geology.arkansas.gov/minerals/industrial/limestone.html">https://www.geology.arkansas.gov/minerals/industrial/limestone.html</a>
- 5. calcaire | GDT Vitrine linguistique Gouvernement du Québec, accessed August 31, 2025,

https://vitrinelinguistique.oqlf.gouv.qc.ca/fiche-gdt/fiche/8878764/calcaire

- 6. Calcaire Wikipédia, accessed August 31, 2025, https://fr.wikipedia.org/wiki/Calcaire
- 7. 12.1 Karst Landscapes and Systems Environmental Geology, accessed August 31, 2025,

https://environmental-geol.pressbooks.tru.ca/chapter/karst-landscapes-and-systems/

- 8. 12.1: Karst Landscapes and Systems Geosciences LibreTexts, accessed August 31, 2025.
  - https://geo.libretexts.org/Bookshelves/Geology/Environmental\_Geology\_(Earle)/12 %3A\_Karst\_and\_Caves/12.01%3A\_Karst\_Landscapes\_and\_Systems
- 9. Limestone | EBSCO Research Starters, accessed August 31, 2025, https://www.ebsco.com/research-starters/chemistry/limestone
- 10. World Karst Aquifer Map WHYMAP, accessed August 31, 2025, <a href="https://www.whymap.org/whymap/EN/Maps\_Data/Wokam/wokam\_inhalt\_en.html">https://www.whymap.org/whymap/EN/Maps\_Data/Wokam/wokam\_inhalt\_en.html</a>
- 11. Karst: Environment and Management of Aquifers The Groundwater Project, accessed August 31, 2025,
  - https://gw-project.org/books/karst-environment-and-management-of-aquifers/
- 12. Geology and landscapes Parc national des calanques, accessed August 31, 2025, <a href="https://www.calanques-parcnational.fr/en/geology-and-landscapes">https://www.calanques-parcnational.fr/en/geology-and-landscapes</a>
- 13. Geology of the Han Cave: a natural wonder shaped by water, accessed August 31, 2025, <a href="https://grotte-de-han.be/en/cave/understanding-the-cave/geology">https://grotte-de-han.be/en/cave/understanding-the-cave/geology</a>
- 14. Limestone | GeoKansas, accessed August 31, 2025, <a href="https://geokansas.ku.edu/limestone">https://geokansas.ku.edu/limestone</a>
- 15. Reliefs Karstiques: Comment l'Eau Façonne le Calcaire?, accessed August 31, 2025, http://odysseedelaterre.fr/reliefs-karstiques-eau-calcaire/
- 16. How Is Limestone Formed And What Is It? Baker Lime, accessed August 31, 2025, <a href="https://www.bakerlime.com/how-limestone-is-formed/">https://www.bakerlime.com/how-limestone-is-formed/</a>
- 17. All You Need to Know About Limestone & How It Formed? World of Stones USA, accessed August 31, 2025, <a href="https://worldofstonesusa.com/blogs/all/what-is-limestone-how-it-formed">https://worldofstonesusa.com/blogs/all/what-is-limestone-how-it-formed</a>
- 18. Limestone | Characteristics, Formation, Texture, Uses, & Facts | Britannica, accessed August 31, 2025, <a href="https://www.britannica.com/science/limestone">https://www.britannica.com/science/limestone</a>
- 19. La formation des roches sédimentaires Assistance scolaire personnalisée et gratuite ASP, accessed August 31, 2025, <a href="https://www.assistancescolaire.com/eleve/5e/svt/reviser-une-notion/la-formation-des-roches-sedimentaires-5sep04/print?print=1&printSheet=1">https://www.assistancescolaire.com/eleve/5e/svt/reviser-une-notion/la-formation-des-roches-sedimentaires-5sep04/print?print=1&printSheet=1</a>
- 20. Une montagne calcaire Parc naturel régional de Chartreuse, accessed August 31, 2025, <a href="https://www.parc-chartreuse.net/content/uploads/2017/12/7\_une\_montagne\_calcaire.pdf">https://www.parc-chartreuse.net/content/uploads/2017/12/7\_une\_montagne\_calcaire.pdf</a>
- 21. Lithification Wikipedia, accessed August 31, 2025, https://en.wikipedia.org/wiki/Lithification
- 22. How does lithification happen in nature, and how long does it take? : r/askscience Reddit, accessed August 31, 2025, <a href="https://www.reddit.com/r/askscience/comments/23c2h4/how\_does\_lithification\_happen\_in\_nature\_and\_how/">https://www.reddit.com/r/askscience/comments/23c2h4/how\_does\_lithification\_happen\_in\_nature\_and\_how/</a>
- 23. Mountain uplift and the Neotectonic Period ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/279576397\_Mountain\_uplift\_and\_the\_Neotectonic\_Period">https://www.researchgate.net/publication/279576397\_Mountain\_uplift\_and\_the\_Neotectonic\_Period</a>
- 24. Massif Central Wikipedia, accessed August 31, 2025,

- https://en.wikipedia.org/wiki/Massif Central
- 25. Geology of the Jura Massif Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/Geology">https://en.wikipedia.org/wiki/Geology</a> of the Jura Massif
- 26. The Karst Landforms and Cycle of Erosion Geomorphology, accessed August 31, 2025, <a href="https://ebooks.inflibnet.ac.in/geop11/chapter/the-karst-landforms-and-cycle-of-erosion/">https://ebooks.inflibnet.ac.in/geop11/chapter/the-karst-landforms-and-cycle-of-erosion/</a>
- 27. Polje | Karst, Sinkholes & Limestone | Britannica, accessed August 31, 2025, <a href="https://www.britannica.com/science/polje">https://www.britannica.com/science/polje</a>
- 28. Proceso Kárstico: Definición & Ejemplos | StudySmarter, accessed August 31, 2025,
  - https://www.studysmarter.es/resumenes/geografia/geociencias/proceso-karstico/
- 29. Geology of Caves USGS Publications Warehouse, accessed August 31, 2025, https://pubs.usgs.gov/gip/7000072/report.pdf
- 30. www2.bgs.ac.uk, accessed August 31, 2025, <a href="https://www2.bgs.ac.uk/mendips/caveskarst/caveform.htm#:~:text=Rainwater%2">https://www2.bgs.ac.uk/mendips/caveskarst/caveform.htm#:~:text=Rainwater%2</a> <a href="https://www2.bgs.ac.uk/mendips/caveskarst/caveform.htm#:~:text=Rainwater%2">https://www2.bgs.ac.uk/mendips/caveskarst/caveform.htm#:~:text=Rainwater%2</a> <a href="https://www2.bgs.ac.uk/mendips/caveskarst/caveform.htm#:~:text=Rainwater%2">https://www2.bgs.ac.uk/mendips/caveskarst/caveform.htm#:~:text=Rainwater%2</a> <a href="https://www2.bgs.ac.uk/mendips/caveskarst/caveform.htm#:~:text=Rainwater%2">https://www2.bgs.ac.uk/mendips/caveskarst/caveform.htm#:~:text=Rainwater%2</a> <a href="https://www.accestarst/caveform.htm#:~:text=Rainwater%2">https://www.accestarst/caveform.htm#:~:text=Rainwater%2</a> <a href="https://www.accestarst/caveform.htm#:~:text=Rainwater%2">https://www.accest
- 31. Karst Formation AustinTexas.gov, accessed August 31, 2025, <a href="https://www.austintexas.gov/sites/default/files/files/Watershed/youth\_education/karst\_lesson\_high\_school.pdf">https://www.austintexas.gov/sites/default/files/files/Watershed/youth\_education/karst\_lesson\_high\_school.pdf</a>
- 32. en.wikipedia.org, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/Karst#:~:text=As%20oxygen%2Drich%20surface%20">https://en.wikipedia.org/wiki/Karst#:~:text=As%20oxygen%2Drich%20surface%20</a> <a href="https://en.wikipedia.org/wiki/Karst#:~:text=As%20oxygen%2Drich%20surface%20">waters,erosion%20within%20the%20limestone%20formation</a>.
- 33. Karst processes and dissolution chemistry | Earth Surface Processes Class Notes Fiveable, accessed August 31, 2025, <a href="https://library.fiveable.me/earth-surface-processes/unit-12/karst-processes-dissolution-chemistry/study-quide/zvQYvYqL5GP0Ssxv">https://library.fiveable.me/earth-surface-processes/unit-12/karst-processes-dissolution-chemistry/study-quide/zvQYvYqL5GP0Ssxv</a>
- 34. Karren (Intro to Geology) Vocab, Definition, Explanations | Fiveable, accessed August 31, 2025,
  - https://library.fiveable.me/key-terms/introduction-geology/karren
- 35. Lapiés | Volcanic, Sedimentary, Igneous | Britannica, accessed August 31, 2025, <a href="https://www.britannica.com/science/lapies">https://www.britannica.com/science/lapies</a>
- 36. Geology Burrenbeo Trust, accessed August 31, 2025, <a href="https://burrenbeo.com/the-burren/natural-heritage/geology/">https://burrenbeo.com/the-burren/natural-heritage/geology/</a>
- 37. Karst Landscapes Caves and Karst (U.S. National Park Service), accessed August 31, 2025, <a href="https://www.nps.gov/subjects/caves/karst-landscapes.htm">https://www.nps.gov/subjects/caves/karst-landscapes.htm</a>
- 38. Karst and Sinkholes WGNHS, accessed August 31, 2025, <a href="https://home.wgnhs.wisc.edu/wisconsin-geology/karst-sinkholes/">https://home.wgnhs.wisc.edu/wisconsin-geology/karst-sinkholes/</a>
- 39. Dolinas o torcas: qué son, formación, tipos y ejemplos Resumen, accessed August 31, 2025, <a href="https://www.geoenciclopedia.com/dolinas-o-torcas-que-son-formacion-tipos-y-eiemplos-1071.html">https://www.geoenciclopedia.com/dolinas-o-torcas-que-son-formacion-tipos-y-eiemplos-1071.html</a>
- 40. Understanding sinkholes and karst British Geological Survey, accessed August 31, 2025, <a href="https://www.bgs.ac.uk/discovering-geology/earth-hazards/sinkholes/">https://www.bgs.ac.uk/discovering-geology/earth-hazards/sinkholes/</a>

- 41. The Study of Karst Landscapes Homeschool.ie, accessed August 31, 2025, https://homeschool.ie/wp-content/uploads/2020/08/LCG-12.pdf
- 42. Polje (Intro to Geology) Vocab, Definition, Explanations | Fiveable, accessed August 31, 2025, <a href="https://library.fiveable.me/key-terms/introduction-geology/polje">https://library.fiveable.me/key-terms/introduction-geology/polje</a>
- 43. THE PLEISTOCENE SEDIMENTARY RECORD OF KARST POLJES IN THE BALKANS (2013 GSA Annual Meeting in Denver Geological Society of America, accessed August 31, 2025,
  - https://gsa.confex.com/gsa/2013AM/webprogram/Paper230079.html
- 44. Limestone caves The Australian Museum, accessed August 31, 2025, <a href="https://australian.museum/learn/minerals/shaping-earth/limestone-caves/">https://australian.museum/learn/minerals/shaping-earth/limestone-caves/</a>
- 45. www.ncei.noaa.gov, accessed August 31, 2025, https://www.ncei.noaa.gov/products/paleoclimatology/speleothem#:~:text=Spele othems%20are%20mineral%20deposits%20formed.to%20determine%20past%2 Oclimate%20conditions.
- 46. Speleothems | Missouri State Parks, accessed August 31, 2025, https://mostateparks.com/page/55118/speleothems
- 47. Speleothem | National Centers for Environmental Information (NCEI), accessed August 31, 2025, https://www.ncei.noaa.gov/products/paleoclimatology/speleothem
- 48. Jura Mountains France.fr, accessed August 31, 2025,
  - https://www.france.fr/en/destination/jura-mountains/
- 49. Deciphering neotectonics from river profile analysis in the karst Jura Mountains (northern Alpine foreland) ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/284019050\_Deciphering\_neotectonics\_from\_river\_profile\_analysis\_in\_the\_karst\_Jura\_Mountains\_northern\_Alpine\_foreland">https://www.researchgate.net/publication/284019050\_Deciphering\_neotectonics\_from\_river\_profile\_analysis\_in\_the\_karst\_Jura\_Mountains\_northern\_Alpine\_foreland</a>
- 50. Structural and Karstic Landscapes of the Joux Valley (Southwestern Jura) ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/342847843\_Structural\_and\_Karstic\_Landscapes">https://www.researchgate.net/publication/342847843\_Structural\_and\_Karstic\_Landscapes</a> of the Joux Valley Southwestern Jura
- 51. 95a calcaires du jurassique moyen du premier plateau du jura NOM DE L'ENTITE N°, accessed August 31, 2025, <a href="https://rhone-mediterranee.eaufrance.fr/sites/sierm/files/content/waterbody\_hydrogeological\_documents/Fiches/95A.pdf">https://rhone-mediterranee.eaufrance.fr/sites/sierm/files/content/waterbody\_hydrogeological\_documents/Fiches/95A.pdf</a>
- 52. KARSTS JURASSIENS Cpepesc, accessed August 31, 2025, https://cpepesc.org/wp-content/uploads/2020/06/Mtg\_KARSTS\_DU\_JURA\_pr\_le\_27-09-03\_diap-2-1.pdf
- 53. The very best of our spectacular landscapes | Jura Mountains, France, accessed August 31, 2025, <a href="https://en.montagnes-du-jura.fr/the-mountains-of-jura-a-nature-destination/top-">https://en.montagnes-du-jura.fr/the-mountains-of-jura-a-nature-destination/top-</a>
  - 7-most-spectacular-landscapes/
- 54. www.researchgate.net, accessed August 31, 2025, https://www.researchgate.net/publication/342847843\_Structural\_and\_Karstic\_Lan\_dscapes\_of\_the\_Joux\_Valley\_Southwestern\_Jura#:~:text=The%20karstic%20land\_scape%20of%20the.which%20contain%20dwarf%20tree%20stands.

- 55. Ice caves as an indicator of winter climate evolution: A case study from the Jura Mountains, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/228492768\_lce\_caves\_as\_an\_indicator\_of\_winter\_climate\_evolution\_A\_case\_study\_from\_the\_Jura\_Mountains">https://www.researchgate.net/publication/228492768\_lce\_caves\_as\_an\_indicator\_of\_winter\_climate\_evolution\_A\_case\_study\_from\_the\_Jura\_Mountains</a>
- 56. TOP 5 caves, caverns and chasms | Jura Mountains, France, accessed August 31, 2025, <a href="https://en.montagnes-du-jura.fr/the-mountains-of-jura-a-nature-destination/sce">https://en.montagnes-du-jura.fr/the-mountains-of-jura-a-nature-destination/sce</a>
  - nery-and-wildlife-of-the-jura-mountains/top-5-caves-caverns-and-chasms-in-the-jura-mountains/
- 57. Impact of global change on karst groundwater mineralization in the Jura Mountains, accessed August 31, 2025, <a href="https://pubmed.ncbi.nlm.nih.gov/26473718/">https://pubmed.ncbi.nlm.nih.gov/26473718/</a>
- 58. Jurassic Karst SNOKARST, accessed August 31, 2025, https://sokarst.org/en/9-observatories-in-france/jurassic-karst-en/
- 59. Jurassic Karst ZAAJ, accessed August 31, 2025, https://zaaj.univ-fcomte.fr/spip.php?article13&lang=en
- 60. Overview of Methods Applied in Karst Hydrogeology KIT AGW, accessed August 31, 2025, <a href="https://www.agw.kit.edu/downloads/Hydro/Karst\_Methods\_Book\_Chapter\_2015.p">https://www.agw.kit.edu/downloads/Hydro/Karst\_Methods\_Book\_Chapter\_2015.p</a> df
- 61. Le karst et les rivières du massif jurassien Saline royale d'Arc-et-Senans, accessed August 31, 2025, <a href="https://www.salineroyale.com/wp-content/uploads/2023/01/Le-karst-et-les-rivieres-du-massif-jurassien-A4.pdf">https://www.salineroyale.com/wp-content/uploads/2023/01/Le-karst-et-les-rivieres-du-massif-jurassien-A4.pdf</a>
- 62. le Karst Jura, accessed August 31, 2025, <a href="http://magnijura.free.fr/geo/Gkarst.html">http://magnijura.free.fr/geo/Gkarst.html</a>
- 63. Karst et karsts régionaux | Observatoire du Risque Inondation de la Sécheresse et du Karst, accessed August 31, 2025, <a href="https://orisk-bfc.fr/karst-et-karsts-r%C3%A9gionaux">https://orisk-bfc.fr/karst-et-karsts-r%C3%A9gionaux</a>
- 64. The Dolomites UNESCO World Heritage Centre, accessed August 31, 2025, <a href="https://whc.unesco.org/en/list/1237/">https://whc.unesco.org/en/list/1237/</a>
- 65. Geology of the Dolomites episodes.org, accessed August 31, 2025, https://www.episodes.org/journal/download\_pdf.php?doi=10.18814/epiiugs/2003/ v26i3/005
- 66. Formation of the Dolomites | History World Heritage Site South Tyrol, accessed August 31, 2025, <a href="https://www.south-tirol.com/culture-holidays-south-tyrol/history/formation-dolomites">https://www.south-tirol.com/culture-holidays-south-tyrol/history/formation-dolomites</a>
- 67. Geology Parco Nazionale Dolomiti Bellunesi, accessed August 31, 2025, <a href="https://www.dolomitipark.it/en/nature-and-history/geology/">https://www.dolomitipark.it/en/nature-and-history/geology/</a>
- 68. Sierra Madre Occidental Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/Sierra Madre Occidental">https://en.wikipedia.org/wiki/Sierra Madre Occidental</a>
- 69. Geology of the Canon De La Huasteca Area in the Sierra Madre Oriental, Nuevo Leon, Mexico Tulane University Journal Publishing, accessed August 31, 2025, <a href="https://journals.tulane.edu/tsgp/article/view/369">https://journals.tulane.edu/tsgp/article/view/369</a>
- 70. Region 9, Sierra Madre Oriental GeoScienceWorld, accessed August 31, 2025, <a href="https://pubs.geoscienceworld.org/books/book/chapter-pdf/3733793/9780813754">https://pubs.geoscienceworld.org/books/book/chapter-pdf/3733793/9780813754</a>

## 673 ch12.pdf

- 71. (PDF) The karst landscapes of Mexico ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/370633360\_The\_karst\_landscapes\_of\_Mexico">https://www.researchgate.net/publication/370633360\_The\_karst\_landscapes\_of\_Mexico</a>
- 72. Sierra Madre Oriental Wikipedia, accessed August 31, 2025, https://en.wikipedia.org/wiki/Sierra Madre Oriental
- 73. How Diverse are the Mountain karst Forests of Mexico? bioRxiv, accessed August 31, 2025, <a href="https://www.biorxiv.org/content/10.1101/2023.03.01.530643v1">https://www.biorxiv.org/content/10.1101/2023.03.01.530643v1</a>
- 74. Sotano de las Golondrinas Association for Mexican Cave Studies, accessed August 31, 2025, <a href="http://www.mexicancaves.org/bul/bul2.pdf">http://www.mexicancaves.org/bul/bul2.pdf</a>
- 75. Cave of Swallows Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/Cave">https://en.wikipedia.org/wiki/Cave</a> of Swallows
- 76. Grutas de García Nuevo León Travel, accessed August 31, 2025, https://www.nuevoleon.travel/grutas-de-garcia/
- 77. Grutas de García Wikipedia, la enciclopedia libre, accessed August 31, 2025, <a href="https://es.wikipedia.org/wiki/Grutas\_de\_Garc%C3%ADa">https://es.wikipedia.org/wiki/Grutas\_de\_Garc%C3%ADa</a>
- 78. Grutas de García Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/Grutas\_de\_Garc%C3%ADa">https://en.wikipedia.org/wiki/Grutas\_de\_Garc%C3%ADa</a>
- 79. Grutas de Garcia Grutas De Garcia, Tour a Grutas De Garcia, accessed August 31, 2025, <a href="https://grutasdegarcia.com/">https://grutasdegarcia.com/</a>
- 80. Tours y actividades Grutas de García Nuevo Leon GetYourGuide, accessed August 31, 2025, https://www.getyourguide.es/grutas-de-garcia-I161641/
- 81. Sierra Gorda Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/Sierra Gorda">https://en.wikipedia.org/wiki/Sierra Gorda</a>
- 82. Sierra Gorda Man and the Biosphere Programme (MAB) UNESCO, accessed August 31, 2025, <a href="https://www.unesco.org/en/mab/sierra-gorda">https://www.unesco.org/en/mab/sierra-gorda</a>
- 83. South China Karst UNESCO World Heritage Centre, accessed August 31, 2025, https://whc.unesco.org/en/list/1248/
- 84. South China Karst Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/South China Karst">https://en.wikipedia.org/wiki/South China Karst</a>
- 85. Guilin Karst IUGS-Geoheritage.org, accessed August 31, 2025, <a href="https://iugs-geoheritage.org/geoheritage.sites/guilin-karst/">https://iugs-geoheritage.org/geoheritage.org/geoheritage.sites/guilin-karst/</a>
- 86. South China Karst Natural World Heritage Sites, accessed August 31, 2025, https://www.naturalworldheritagesites.org/sites/south-china-karst/
- 87. South China Karst UNESCO World Heritage, accessed August 31, 2025, <a href="https://www.travelchinaguide.com/attraction/worldheritages/southkarst.htm">https://www.travelchinaguide.com/attraction/worldheritages/southkarst.htm</a>
- 88. Trang An Landscape Complex UNESCO World Heritage Centre, accessed August 31, 2025, <a href="https://whc.unesco.org/en/list/1438/">https://whc.unesco.org/en/list/1438/</a>
- 89. Ancient Villages of Northern Syria UNESCO World Heritage Centre, accessed August 31, 2025, <a href="https://whc.unesco.org/en/list/1348/">https://whc.unesco.org/en/list/1348/</a>
- 90. (PDF) Karst evolution of the Nullarbor Plain, Australia ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/246599692\_Karst\_evolution\_of\_the\_Nullarbor Plain Australia">https://www.researchgate.net/publication/246599692\_Karst\_evolution\_of\_the\_Nullarbor Plain Australia</a>
- 91. Nullarbor Plain Wikipedia, accessed August 31, 2025,

- https://en.wikipedia.org/wiki/Nullarbor Plain
- 92. Origin and karst geomorphological significance of the enigmatic Australian Nullarbor Plain blowholes Ovid, accessed August 31, 2025, <a href="https://www.ovid.com/journals/espl/pdf/10.1002/esp.2233~origin-and-karst-geomorphological-significance-of-the">https://www.ovid.com/journals/espl/pdf/10.1002/esp.2233~origin-and-karst-geomorphological-significance-of-the</a>
- 93. Nullarbor Karst The Western Australian Speleological Group, accessed August 31, 2025, https://wasg.org.au/wa-caving-areas/nullarbor/nullarbor-karst
- 94. Tsingy of Madagascar What is it & How is it Formed? Aqua Firma, accessed August 31, 2025, <a href="https://www.aqua-firma.com/tsingy-madagascar">https://www.aqua-firma.com/tsingy-madagascar</a>
- 95. Tsingy de Bemaraha National Park Wikipedia, accessed August 31, 2025, https://en.wikipedia.org/wiki/Tsingy de Bemaraha National Park
- 96. Tsingy de Bemaraha Atlas Obscura's Where to Wander, accessed August 31, 2025, https://wheretowander2024.atlasobscura.com/location/madagascar
- 97. Tsingy De Bemaraha Strict Nature Reserve Madagascar African World Heritage Sites, accessed August 31, 2025, <a href="https://www.africanworldheritagesites.org/natural-places/strange-worlds/tsingy-de-bemaraha-madagascar.html">https://www.africanworldheritagesites.org/natural-places/strange-worlds/tsingy-de-bemaraha-madagascar.html</a>
- 98. Biodiversity in Karst Landscapes: Introduction to the Special Issue MDPI, accessed August 31, 2025, https://www.mdpi.com/1424-2818/15/3/408
- 99. Biodiversity in Karst Landscapes: Introduction to the Special Issue ResearchGate, accessed August 31, 2025,
  <a href="https://www.researchgate.net/publication/369227137">https://www.researchgate.net/publication/369227137</a> Biodiversity in Karst Lands
  <a href="mailto:capes">capes</a> Introduction to the Special Issue
- 100. Special Issue: Biodiversity in Karst Landscapes MDPI, accessed August 31, 2025, https://www.mdpi.com/journal/diversity/special\_issues/Karst\_Landscapes
- 101. Karst Plants Diversity, Evolution, Taxonomy and Conservation Frontiers, accessed August 31, 2025, <a href="https://www.frontiersin.org/research-topics/59976/karst-plants-diversity-evolution-taxonomy-and-conservation">https://www.frontiersin.org/research-topics/59976/karst-plants-diversity-evolution-taxonomy-and-conservation</a>
- 102. Calcicole Wikipedia, accessed August 31, 2025, https://en.wikipedia.org/wiki/Calcicole
- 103. Immobilization of tissue iron on calcareous soil: differences between calcicole and calcifuge plants | Lund University, accessed August 31, 2025, <a href="https://www.lunduniversity.lu.se/lup/publication/000d3a98-ae46-42be-acfa-3948-98e50dff">https://www.lunduniversity.lu.se/lup/publication/000d3a98-ae46-42be-acfa-3948-98e50dff</a>
- 104. Soluble Inorganic Tissue Phosphorus and Calcicole—Calcifuge Behaviour of Plants | Annals of Botany | Oxford Academic, accessed August 31, 2025, <a href="https://academic.oup.com/aob/article/94/3/427/167351">https://academic.oup.com/aob/article/94/3/427/167351</a>
- 105. en.wikipedia.org, accessed August 31, 2025, https://en.wikipedia.org/wiki/Calcicole#:~:text=contemporary%20gene%20flow.-, Physiological%20adaptations.tip%20cells%20and%20epidermal%20bladders.
- 106. Troglofauna Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/Troglofauna">https://en.wikipedia.org/wiki/Troglofauna</a>
- 107. Cave life: creatures that lurk in the shadows | Natural History Museum, accessed August 31, 2025,

- https://www.nhm.ac.uk/discover/creatures-that-lurk-in-caves.html
- 108. Extraordinary and Endangered Cave Dwelling-Animals, accessed August 31, 2025, https://www.mossy.earth/rewilding-knowledge/cave-dwelling-animals
- 109. List of troglobites Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/List\_of-troglobites">https://en.wikipedia.org/wiki/List\_of-troglobites</a>
- 110. Subterranean (Hypogean) Habitats in Karst and Their Fauna | Request PDF ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/309306658\_Subterranean\_Hypogean\_Habitats\_in\_Karst\_and\_Their\_Fauna">https://www.researchgate.net/publication/309306658\_Subterranean\_Hypogean\_Habitats\_in\_Karst\_and\_Their\_Fauna</a>
- 111. An Introduction to Some Cave Fauna of Malaysia and Thailand Digital Commons @ USF University of South Florida, accessed August 31, 2025, <a href="https://digitalcommons.usf.edu/cgi/viewcontent.cgi?article=1248&context=kip\_article="https://digitalcommons.usf.edu/cgi/viewcontent.cgi?article=1248&context=kip\_article="https://digitalcommons.usf.edu/cgi/viewcontent.cgi?article=1248&context=kip\_article="https://digitalcommons.usf.edu/cgi/viewcontent.cgi?article=1248&context=kip\_article="https://digitalcommons.usf.edu/cgi/viewcontent.cgi">https://digitalcommons.usf.edu/cgi/viewcontent.cgi?article=1248&context=kip\_article=1248&context=ki
- 112. Troglofauna (Subterranean Fauna) Bennelongia Environmental Consultants, accessed August 31, 2025,
  - https://www.bennelongia.com.au/services/subterranean-fauna/troglofauna/
- 113. Troglofauna in the vadose zone: comparison of scraping and trapping results and sampling adequacy Subterranean Biology, accessed August 31, 2025, <a href="https://subtbiol.pensoft.net/article/1318/">https://subtbiol.pensoft.net/article/1318/</a>
- 114. Troglofauna: Animals of the Caves YouTube, accessed August 31, 2025, <a href="https://www.youtube.com/watch?v=XOUvj6wufl8">https://www.youtube.com/watch?v=XOUvj6wufl8</a>
- 115. Limestone, accessed August 31, 2025, https://nora.nerc.ac.uk/id/eprint/534436/1/mpf\_limestone.pdf
- 116. Derbyshire and Derby Minerals Local Plan 2022 2038 ..., accessed August 31, 2025,
  - https://www.derbyshire.gov.uk/site-elements/documents/pdf/council/have-your-say/minerals-local-plan/7.2a-industrial-limestone-background-paper.pdf
- 117. The Versatile Rock: Exploring Limestone Uses in Various Industries, accessed August 31, 2025,
  - https://pontotocsandandstone.com/limestone-in-different-industries/
- 118. Socio-Economic Impacts of Quarrying and Processing of Limestone in Ewekoro, Nigeria EA Journals, accessed August 31, 2025, <a href="https://www.eajournals.org/wp-content/uploads/Communities-Company-Relation-in-Limestone-Mining-Region-of-Southwest-Nigeria.pdf">https://www.eajournals.org/wp-content/uploads/Communities-Company-Relation-in-Limestone-Mining-Region-of-Southwest-Nigeria.pdf</a>
- ISSN 2319-829X Humanities and Social Science Studies, Vol. 12(1) Natural and socio-economic impacts of stone quarrying - HSSS Journal, accessed August 31, 2025,
  - https://hsssjournal.com/wp-content/uploads/2024/06/11.-2023-Vol-12-Issue-1-Samir-Pramanik-Bela-Das.pdf
- 120. The Social, Economic and Environmental Benefits and Drawbacks of Limestone Quarrying GCSE Chemistry (Triple) WJEC Revision Study Rocket, accessed August 31, 2025,
  - https://studyrocket.co.uk/revision/gcse-chemistry-triple-wjec/limestone/the-social-economic-and-environmental-benefits-and-drawbacks-of-limestone-q
- 121. Risks, challenges and socioeconomic impacts of quarrying on rural livelihoods

- in Tanzania, accessed August 31, 2025, <a href="https://www.tandfonline.com/doi/full/10.1080/23311886.2025.2489040">https://www.tandfonline.com/doi/full/10.1080/23311886.2025.2489040</a>
- 122. Jamaica ACP-EU Development Minerals Programme, accessed August 31, 2025, <a href="https://developmentminerals.org/jamaica/">https://developmentminerals.org/jamaica/</a>
- Jamaica's Mining and Quarrying Sector Performance 2023, accessed August 31, 2025,

  https://mgd.gov.im/wp.content/upleads/2024/10/Jamaica\_Mining\_Sector\_Performance
  - https://mgd.gov.jm/wp-content/uploads/2024/10/Jamaica\_Mining\_Sector\_Performance\_2023.pdf
- 124. Mining and Quarrying Sector Plan, accessed August 31, 2025, https://mgd.gov.jm/wp-content/uploads/2022/09/Mining-and-Quarrying-Sector-Plan.pdf
- 125. Limestone Quarry Waste Promotes the Growth of Two Native Woody Angiosperms - Frontiers, accessed August 31, 2025, <a href="https://www.frontiersin.org/journals/ecology-and-evolution/articles/10.3389/fevo.2">https://www.frontiersin.org/journals/ecology-and-evolution/articles/10.3389/fevo.2 021.637833/full</a>
- 126. www.whymap.org, accessed August 31, 2025, https://www.whymap.org/whymap/EN/Maps\_Data/Wokam/wokam\_inhalt\_en.html #:~:text=ln%20many%20countries%20and%20regions,or%20predominantly%20 on%20karst%20groundwater.
- 127. Living with Karst The Watershed Association, accessed August 31, 2025, <a href="https://watershedassociation.org/impactareas/education-and-outreach/karst/">https://watershedassociation.org/impactareas/education-and-outreach/karst/</a>
- 128. Water Resources Karst Waters Institute, accessed August 31, 2025, https://karstwaters.org/educational-resources/water-resources/
- 129. Environmental issues in karst regions | Earth Surface Processes Class Notes Fiveable, accessed August 31, 2025, <a href="https://library.fiveable.me/earth-surface-processes/unit-12/environmental-issues-karst-regions/study-guide/O9dmGWKIRHqpSmhj">https://library.fiveable.me/earth-surface-processes/unit-12/environmental-issues-karst-regions/study-guide/O9dmGWKIRHqpSmhj</a>
- 130. Lesson 5: What Makes Karst Sensitive and Vulnerable to ... Gov.bc.ca, accessed August 31, 2025, <a href="https://www2.gov.bc.ca/gov/content/industry/forestry/managing-our-forest-resources/managed-resource-features/18190/18194">https://www2.gov.bc.ca/gov/content/industry/forestry/managing-our-forest-resources/managed-resource-features/18190/18194</a>
- 131. Living on Karst Cave Conservancy of the Virginias, accessed August 31, 2025, <a href="https://caveconservancyofvirginia.org/karst/">https://caveconservancyofvirginia.org/karst/</a>
- 132. Protecting water in karst regions | Minnesota Pollution Control Agency, accessed August 31, 2025, <a href="https://www.pca.state.mn.us/air-water-land-climate/protecting-water-in-karst-regions">https://www.pca.state.mn.us/air-water-land-climate/protecting-water-in-karst-regions</a>
- 133. Ground Water Investigations in Karst Areas Minnesota Pollution Control Agency, accessed August 31, 2025, <a href="https://www.pca.state.mn.us/sites/default/files/c-prp4-09.pdf">https://www.pca.state.mn.us/sites/default/files/c-prp4-09.pdf</a>
- 134. Edwards Aquifer Protection Plans Texas Commission on Environmental Quality, accessed August 31, 2025, https://www.tceq.texas.gov/permitting/eapp/apps.html
- 135. Edwards Aquifer Protection Plan, accessed August 31, 2025, <a href="https://aquiferalliance.org/Library/LibraryFiles/Resources/Edwards Aquifer Protection">https://aquiferalliance.org/Library/LibraryFiles/Resources/Edwards Aquifer Protection</a>

## tion Plan.pdf

- 136. Edwards Aquifer Protection Program Texas Commission on Environmental Quality, accessed August 31, 2025, <a href="https://www.tceq.texas.gov/permitting/eapp">https://www.tceq.texas.gov/permitting/eapp</a>
- 137. Edwards Aquifer Monitoring and Research Plan Ecosystem Planning & Restoration, accessed August 31, 2025, <a href="https://www.eprusa.net/projects/edwards-aquifer-monitoring-and-research-plan">https://www.eprusa.net/projects/edwards-aquifer-monitoring-and-research-plan</a>
- 138. Managing The Edwards Aquifer, accessed August 31, 2025, <a href="https://www.edwardsaquifer.org/groundwater-users/manage/">https://www.edwardsaquifer.org/groundwater-users/manage/</a>
- 139. Exploring the History of Limestone in Architecture and Its Usage Over Time | Outside Stone, accessed August 31, 2025, <a href="https://outsidestone.com/exploring-the-history-of-limestone-in-architecture-and-its-usage-over-time/">https://outsidestone.com/exploring-the-history-of-limestone-in-architecture-and-its-usage-over-time/</a>
- 140. Limestone Architextures, accessed August 31, 2025, https://architextures.org/textures/2164
- 141. Famous Buildings That Are Made Using Limestone Regatta Universal Exports, accessed August 31, 2025, <a href="https://www.regattaexports.com/famous-buildings-that-are-made-using-limestone/">https://www.regattaexports.com/famous-buildings-that-are-made-using-limestone/</a>
- 142. Limestone and Recreation Internet Geography, accessed August 31, 2025, <a href="https://www.internetgeography.net/topics/limestone-and-recreation/">https://www.internetgeography.net/topics/limestone-and-recreation/</a>
- 143. What Are Some of the Best Uses for Limestone? Stoneco, accessed August 31, 2025, <a href="https://www.stoneco.net/about-us/press/limestone-uses">https://www.stoneco.net/about-us/press/limestone-uses</a>
- 144. Caving Washington and Lee University, accessed August 31, 2025, <a href="https://my.wlu.edu/outing-club/outdoor-activities/caving">https://my.wlu.edu/outing-club/outdoor-activities/caving</a>
- 145. How caves form | Caves and karst | Foundations of the Mendips - British Geological Survey, accessed August 31, 2025, https://www2.bgs.ac.uk/mendips/caveskarst/caveform.htm
- 146. Cave Impacts National Park Service, accessed August 31, 2025, <a href="https://www.nps.gov/wica/learn/nature/cave-impacts.htm">https://www.nps.gov/wica/learn/nature/cave-impacts.htm</a>
- 147. Cave Life Great Basin National Park (U.S. National Park Service), accessed August 31, 2025, <a href="https://www.nps.gov/grba/learn/nature/cave-life.htm">https://www.nps.gov/grba/learn/nature/cave-life.htm</a>
- 148. Limestone Case Study Malham, The Yorkshire Dales Internet Geography, accessed August 31, 2025, <a href="https://www.internetgeography.net/topics/limestone-case-study-malham-the-yorkshire-dales/">https://www.internetgeography.net/topics/limestone-case-study-malham-the-yorkshire-dales/</a>
- 149. Karst Environments: Problems, Management, Human Impacts, and Sustainability: An Introduction to the Special Issue ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/233579447\_Karst\_Environments\_Problems\_Management\_Human\_Impacts\_and\_Sustainability\_An\_Introduction\_to\_the\_Special\_Issue">https://www.researchgate.net/publication/233579447\_Karst\_Environments\_Problems\_Management\_Human\_Impacts\_and\_Sustainability\_An\_Introduction\_to\_the\_Special\_Issue</a>
- 150. Foundation Engineering Problems and Hazards in Karst Terranes Maryland Geological Survey, accessed August 31, 2025, <a href="http://www.mgs.md.gov/geology/geohazards/engineering\_problems\_in\_karst.htm">http://www.mgs.md.gov/geology/geohazards/engineering\_problems\_in\_karst.htm</a>

- 151. Top 16Focus Areas Global Geoparks Network, accessed August 31, 2025, <a href="https://www.globalgeoparksnetwork.org/sites/default/files/2024-03/GGN\_MAP\_AFISA 2023-2024\_XROMATA\_TELIKO\_1.pdf">https://www.globalgeoparksnetwork.org/sites/default/files/2024-03/GGN\_MAP\_AFISA 2023-2024\_XROMATA\_TELIKO\_1.pdf</a>
- 152. UNESCO Global Geoparks Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/UNESCO Global Geoparks">https://en.wikipedia.org/wiki/UNESCO Global Geoparks</a>
- 153. Shilin UNESCO Global Geopark, accessed August 31, 2025, <a href="https://www.unesco.org/en/iggp/shilin-unesco-global-geopark">https://www.unesco.org/en/iggp/shilin-unesco-global-geopark</a>
- 154. Tourism in the Dales Yorkshire Dales National Park, accessed August 31, 2025.
  - https://www.yorkshiredales.org.uk/park-authority/living-and-working/tourism-in-the-dales/
- 155. yorkshire dales national park management plan 2025-2030, accessed August 31, 2025,
  - https://www.yorkshiredales.org.uk/wp-content/uploads/sites/13/2025/01/Draft-SA-Report-Nov-2024.pdf
- 156. Management Plan Yorkshire Dales National Park, accessed August 31, 2025, https://www.yorkshiredales.org.uk/wp-content/uploads/sites/13/2025/03/YDNP-M anagement-Plan-2019-24-updated-Nov-2021.pdf
- 157. Progress on National Park plans, accessed August 31, 2025, https://www.yorkshiredales.org.uk/progress-on-national-park-plans/
- 158. ingeoexpert.com, accessed August 31, 2025,

  <a href="https://ingeoexpert.com/2019/09/23/que-es-el-relieve-karstico-como-se-forma/#:~:text=La%20disoluci%C3%B3n%20k%C3%A1rstica%20se%20debe,c%C3%A1lcico%2C%20que%20es%20altamente%20soluble.">https://ingeoexpert.com/2019/09/23/que-es-el-relieve-karstico-como-se-forma/#:~:text=La%20disoluci%C3%B3n%20k%C3%A1rstica%20se%20debe,c%C3%A1lcico%2C%20que%20es%20altamente%20soluble.</a>
- 159. (PDF) Procesos de karstificación ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/305474239">https://www.researchgate.net/publication/305474239</a> Procesos de karstificación
- 160. Dolina Geología JovaGeology, accessed August 31, 2025, <a href="https://www.jovageology.com/2021/07/dolina-geologia.html">https://www.jovageology.com/2021/07/dolina-geologia.html</a>
- 161. Márton Veress: Karst Environments: Karren Formation in High Mountains. Springer The Distant Reader, accessed August 31, 2025, <a href="https://distantreader.org/stacks/journals/hungeobull/hungeobull-3095.pdf">https://distantreader.org/stacks/journals/hungeobull-3095.pdf</a>
- 162. 10 Most famous caves in the world Oxalis Adventure, accessed August 31, 2025, https://oxalisadventure.com/10-most-famous-caves-in-the-world/
- 163. Les 10 Plus Belles Grottes du Monde | L'Odyssée de la Terre, accessed August 31, 2025, <a href="https://odysseedelaterre.fr/plus-belles-grottes-du-monde/">https://odysseedelaterre.fr/plus-belles-grottes-du-monde/</a>
- 164. Voyage en immersion dans les régions karstiques du nord Vietnam Tamera, accessed August 31, 2025, <a href="https://www.tamera.fr/blog/vietnam/voyage-en-immersion-dans-les-regions-karstiques-du-nord-vietnam">https://www.tamera.fr/blog/vietnam/voyage-en-immersion-dans-les-regions-karstiques-du-nord-vietnam</a>
- 165. les grands massifs karstiques de la france EFS, accessed August 31, 2025, https://efs.ffspeleo.fr/images/di les grands massifs karstiques de france.pdf
- 166. Les formations calcaires et le karst | L'eau dans le bassin Rhône-Méditerranée, accessed August 31, 2025,
  - https://www.rhone-mediterranee.eaufrance.fr/milieux-aquatiqueseaux-souterrain

- escontexte-hydrogeologique/les-formations-calcaires-et-le-karst
- 167. CALIZA Fdminerals, minerales de colección, accessed August 31, 2025, <a href="https://www.fdminerals.es/2018/07/24/caliza/">https://www.fdminerals.es/2018/07/24/caliza/</a>
- 168. 11 Tipos de piedra Caliza y Arenisca Pulycort, accessed August 31, 2025, <a href="https://www.pulycort.com/piedras-calizas-areniscas/tipos-de-caliza-y-arenisca.h">https://www.pulycort.com/piedras-calizas-areniscas/tipos-de-caliza-y-arenisca.h</a> tml
- 169. Etapa 8: Historia de la construcción Geo-Sports, accessed August 31, 2025, <a href="https://www.geo-sports.org/es/2024/etapa-8-historia-de-la-construccion/">https://www.geo-sports.org/es/2024/etapa-8-historia-de-la-construccion/</a>
- 170. Piedras calizas y | conversa Wix.com, accessed August 31, 2025, https://conversaonline.wixsite.com/conversa/piedras-calizas-y
- 171. Compilation and Production of a Karst map of Mexico ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/252157431\_Compilation\_and\_Production\_nof-a-Karst map-of-Mexico">https://www.researchgate.net/publication/252157431\_Compilation\_and\_Production\_nof-a-Karst map-of-Mexico</a>
- 172. Limestone ecosystems: Our work Department of Conservation, accessed August 31, 2025, <a href="https://www.doc.govt.nz/our-work/limestone-ecosystems/">https://www.doc.govt.nz/our-work/limestone-ecosystems/</a>
- 173. The geomicrobiology of limestone, sulfuric acid speleogenetic, and volcanic caves: basic concepts and future perspectives PubMed Central, accessed August 31, 2025, <a href="https://pmc.ncbi.nlm.nih.gov/articles/PMC10987966/">https://pmc.ncbi.nlm.nih.gov/articles/PMC10987966/</a>
- 174. Sierra de Grazalema Man and the Biosphere Programme (MAB) UNESCO, accessed August 31, 2025, <a href="https://www.unesco.org/en/mab/sierra-de-grazalema">https://www.unesco.org/en/mab/sierra-de-grazalema</a>
- 175. The Northern Limestone Conservation Landscape TRUNG TÂM BÁO TÔN THIÊN NHIÊN VÀ PHÁT TRIỂN, accessed August 31, 2025, <a href="https://ccd.org.vn/en/protected-areas-governance/the-northern-limestone-conservation-landscape/">https://ccd.org.vn/en/protected-areas-governance/the-northern-limestone-conservation-landscape/</a>
- 176. (PDF) KARST AND ECOSYSTEM SERVICE IN THE LARGE LIMESTONES MASSIF(T 3 -J 1) OF MALI I THATË-RAKICKA, SOUTHEASTERN ALBANIA ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/363845327\_KARST\_AND\_ECOSYSTEM\_SERVICE\_IN\_THE\_LARGE\_LIMESTONES\_MASSIFT\_3 -J 1 OF MALI I THATE-RAKICKA SOUTHEASTERN ALBANIA</a>
- 177. Karst and caves | Land and Soil Environment and Heritage NSW Government, accessed August 31, 2025, <a href="https://www.environment.nsw.gov.au/topics/land-and-soil/geodiversity/karst-and-caves">https://www.environment.nsw.gov.au/topics/land-and-soil/geodiversity/karst-and-caves</a>
- 178. The Significance of the Calcicolous Habit Sci-Hub, accessed August 31, 2025, <a href="https://dacemirror.sci-hub.se/journal-article/aa95b11fddde8f542035d07d1a5e3a4">https://dacemirror.sci-hub.se/journal-article/aa95b11fddde8f542035d07d1a5e3a4</a> e/salisbury1920.pdf
- 179. (PDF) The secrets of calcicole species revealed ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/349581949\_The\_secrets\_of\_calcicole\_species\_revealed">https://www.researchgate.net/publication/349581949\_The\_secrets\_of\_calcicole\_species\_revealed</a>
- secrets of calcicole species revealed | Journal of Experimental Botany -Oxford Academic, accessed August 31, 2025, <a href="https://academic.oup.com/jxb/article/72/4/968/6149237">https://academic.oup.com/jxb/article/72/4/968/6149237</a>

- 181. Cave Creatures Vocabulary Texas Parks & Wildlife Department, accessed August 31, 2025,
  - https://tpwd.texas.gov/education/resources/keep-texas-wild/cave-creatures/cave-creatures-vocabulary
- 182. Fauna of Northwest caves Oregon Grotto, accessed August 31, 2025, <a href="https://www.oregongrotto.com/fauna-of-northwest-caves/">https://www.oregongrotto.com/fauna-of-northwest-caves/</a>
- 183. Fauna Turismo de Grazalema, accessed August 31, 2025, <a href="https://turismo.grazalema.es/index.php/rutas/fauna/157-fauna">https://turismo.grazalema.es/index.php/rutas/fauna/157-fauna</a>
- 184. Macizo de Larra-Belagua Wikipedia, la enciclopedia libre, accessed August 31, 2025, <a href="https://es.wikipedia.org/wiki/Macizo de Larra-Belagua">https://es.wikipedia.org/wiki/Macizo de Larra-Belagua</a>
- 185. Los prados de piedra caliza del Gran Sitio de Solutré, accessed August 31, 2025,
  - https://rochedesolutre.com/es/contemplar-los-paisajes/un-mosaico-de-paisajes/c%C3%A9spedes/
- 186. Contemplar reserva natural del humedal de Van Long Nhan Dan Online, accessed August 31, 2025,
  - https://es.nhandan.vn/contemplar-reserva-natural-del-humedal-de-van-long-post62796.html
- 187. Sierra de Orihuela, accessed August 31, 2025, https://deportes.orihuela.es/Static.php?c=144
- 188. Éboulis calcaires et de schistes calcaires des étages montagnard à alpin Dreal PACA, accessed August 31, 2025, <a href="https://www.paca.developpement-durable.gouv.fr/IMG/pdf/1\_H8120\_cle2ee57c.pdf">https://www.paca.developpement-durable.gouv.fr/IMG/pdf/1\_H8120\_cle2ee57c.pdf</a>
- 189. Massifs calcaires: Définition & Formation StudySmarter, accessed August 31, 2025.
  - https://www.studysmarter.fr/resumes/geographie/geosciences/massifs-calcaires/
- 190. Modeling of groundwater processes in a karstic aquifer of Sierra Madre Oriental, Mexico | Request PDF ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/325824727\_Modeling\_of\_groundwater\_processes">https://www.researchgate.net/publication/325824727\_Modeling\_of\_groundwater\_processes</a> in a karstic aquifer of Sierra Madre Oriental Mexico
- 191. What is the Jura? Geology explained | Geopark Schwäbische Alb, accessed August 31, 2025,
  - https://www.geopark-alb.de/en/understand-geopark/geology-explained/detail/what-is-the-jura
- 192. Geoparks the basis of integrated conservation of ecosystems, culture and development of ecotourism in the Kyrgyz Tian-Shan Тянь-Шаньское Геологическое Общество, accessed August 31, 2025, <a href="https://www.geotianshan.org/en/projects/geopark-of-kyrgyzstan/">https://www.geotianshan.org/en/projects/geopark-of-kyrgyzstan/</a>
- 193. Calcifuges and Calcicoles Edinburgh Garden School, accessed August 31, 2025,
  - https://www.edinburghgardenschool.com/2015/05/calcifuges-and-calcicoles/
- 194. Subterranean fauna Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/Subterranean fauna">https://en.wikipedia.org/wiki/Subterranean fauna</a>
- 195. Sierra Madre Oriental PeakVisor, accessed August 31, 2025,

- https://peakvisor.com/range/sierra-madre-oriental.html
- 196. Geology Réserve Naturelle Nationale de la Haute Chaîne du Jura, accessed August 31, 2025,
  - https://www.rnn-hautechainedujura.fr/en/natural-heritage/geology/
- 197. (PDF) A case study of the life cycle impact of limestone quarrying on the environment, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/347287970\_A\_case\_study\_of\_the\_life\_cycle\_impact\_of\_limestone\_quarrying\_on\_the\_environment">https://www.researchgate.net/publication/347287970\_A\_case\_study\_of\_the\_life\_cycle\_impact\_of\_limestone\_quarrying\_on\_the\_environment</a>
- 198. Karst Aquifers | U.S. Geological Survey USGS.gov, accessed August 31, 2025, https://www.usgs.gov/mission-areas/water-resources/science/karst-aquifers
- 199. (PDF) The Impact of Tourism Activity on Coastal Biodiversity: A Case Study at Praia da Cova Redonda (Algarve—Portugal) ResearchGate, accessed August 31, 2025.
  - https://www.researchgate.net/publication/344636673 The Impact of Tourism Activity on Coastal Biodiversity A Case Study at Praia da Cova Redonda Algarve-Portugal
- 200. The Impact of Tourism Activity on Coastal Biodiversity: A Case Study at Praia da Cova Redonda (Algarve—Portugal) MDPI, accessed August 31, 2025, https://www.mdpi.com/2076-3298/7/10/88
- 201. Case Studies | EurolithosCommunity, accessed August 31, 2025, <a href="https://www.eurolithos.org/casestudies">https://www.eurolithos.org/casestudies</a>
- 202. SOUTH CHINA KARST World Heritage Datasheet, accessed August 31, 2025, <a href="http://world-heritage-datasheets.unep-wcmc.org/datasheet/output/site/south-china-karst/">http://world-heritage-datasheets.unep-wcmc.org/datasheet/output/site/south-china-karst/</a>
- 203. Sierra Madre Oriental Pine-Oak Forests | One Earth, accessed August 31, 2025.
  - https://www.oneearth.org/ecoregions/sierra-madre-oriental-pine-oak-forests/
- 204. MEXICO THE 5 MOST BEAUTIFUL CENOTES OF THE YUCATAN | TRAVEL BLOG, accessed August 31, 2025,
  - https://www.elovoyage.com/en/Mexico%3A-The-5-Most-Beautiful-Cenotes-in-the-Yucatan/
- 205. Jura Mountains Wikipedia, accessed August 31, 2025, <a href="https://en.wikipedia.org/wiki/Jura\_Mountains">https://en.wikipedia.org/wiki/Jura\_Mountains</a>
- 206. Speleology caves' exploring Polish Jura, accessed August 31, 2025, <a href="https://polish-jura.com/speleology-caves-exploring/">https://polish-jura.com/speleology-caves-exploring/</a>
- 207. Swabian Alp | Hiking, Skiing & Nature Britannica, accessed August 31, 2025, https://www.britannica.com/place/Swabian-Alp
- 208. Karst path of the big wood and Maëva cave Montagnes du Jura, accessed August 31, 2025,
  - https://en.montagnes-du-jura.fr/hiking/sentier-karstique-du-grand-bois-et-grotte -maeva-3
- 209. Karst Features and Aquifers in Minnesota, accessed August 31, 2025, <a href="https://minnesotaprotectourwater.com/karst-features-and-aquifers-in-minnesota">https://minnesotaprotectourwater.com/karst-features-and-aquifers-in-minnesotal</a>
- 210. ECOTOURISM IN THE STATE FOREST KARST OF PUERTO RICO National

- Speleological Society, accessed August 31, 2025, <a href="https://legacy.caves.org/pub/journal/PDF/v76/cave-76-01-30r.pdf">https://legacy.caves.org/pub/journal/PDF/v76/cave-76-01-30r.pdf</a>
- 211. Ecotourism in the State Forest Karst of Puerto Rico | Request PDF ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/publication/276030890\_Ecotourism\_in\_the\_State\_F">https://www.researchgate.net/publication/276030890\_Ecotourism\_in\_the\_State\_F</a> orest Karst of Puerto Rico
- 212. Green Tourism Planning for Coastal Development in Gunungsewu Geopark, Indonesia | International Journal of Geoinformatics Public Knowledge Project, accessed August 31, 2025, https://journals.sfu.ca/ijg/index.php/journal/article/view/2701
- 213. The Dolomites | World Heritage Outlook IUCN, accessed August 31, 2025, <a href="https://worldheritageoutlook.iucn.org/explore-sites/dolomites">https://worldheritageoutlook.iucn.org/explore-sites/dolomites</a>
- 214. The Dolomites 2017 Conservation Outlook Assessment (archived) Fondazione Dolomiti Unesco, accessed August 31, 2025, <a href="https://www.dolomitiunesco.info/wp-content/uploads/2023/03/The-Dolomites-2017-COA-en.pdf">https://www.dolomitiunesco.info/wp-content/uploads/2023/03/The-Dolomites-2017-COA-en.pdf</a>
- 215. IUCN Red List Habitat Classification BASIN Field Manual, accessed August 31, 2025,
  - https://docs.basin.global/appendix/land-cover-classification-and-ecosystem-typologies/ecosystem-classification-frameworks/iucn-red-list-habitat-classification
- 216. IUCN Habitats Classification Scheme, accessed August 31, 2025, <a href="https://nc.iucnredlist.org/redlist/content/attachment\_files/dec\_2012\_guidance\_habitats\_classification\_scheme.pdf">https://nc.iucnredlist.org/redlist/content/attachment\_files/dec\_2012\_guidance\_habitats\_classification\_scheme.pdf</a>
- 217. Habitats Classification Scheme (Version 3.1) IUCN Red List of Threatened Species, accessed August 31, 2025, https://www.iucnredlist.org/resources/habitat-classification-scheme
- 218. www.oneearth.org, accessed August 31, 2025, <a href="https://www.oneearth.org/ecoregions/nullarbor-plains-xeric-shrublands/#:~:text=Roughly%20630%20plant%20species%20have.blue%20bonnet%20parrot%20occur%20here.">https://www.oneearth.org/ecoregions/nullarbor-plains-xeric-shrublands/#:~:text=Roughly%20630%20plant%20species%20have.blue%20bonnet%20parrot%20occur%20here.</a>
- 219. Sótano de las Golondrinas en la Huasteca Potosina Travesías, accessed August 31, 2025, <a href="https://www.travesiasdigital.com/destinos/mexico/sotano-de-las-golondrinas-en-la-huasteca-potosina/">https://www.travesiasdigital.com/destinos/mexico/sotano-de-las-golondrinas-en-la-huasteca-potosina/</a>
- 220. Sótano de las Golondrinas area with elevation contour layers. Photo by the author. ResearchGate, accessed August 31, 2025, <a href="https://www.researchgate.net/figure/Sotano-de-las-Golondrinas-area-with-elevation-contour-layers-Photo-by-the-author-fig4">https://www.researchgate.net/figure/Sotano-de-las-Golondrinas-area-with-elevation-contour-layers-Photo-by-the-author-fig4</a> 352736520
- 221. Sotano de las Golondrinas ropewiki, accessed August 31, 2025, <a href="https://ropewiki.com/Sotano">https://ropewiki.com/Sotano</a> de las Golondrinas
- 222. Market Survey Report on Mining & Exportation of Limestone High Commission Of India, Kingston, Jamaica, accessed August 31, 2025, <a href="https://hcikingston.gov.in/public\_files/assets/pdf/Market%20Survey%20Report%2">https://hcikingston.gov.in/public\_files/assets/pdf/Market%20Survey%20Report%2</a> <a href="https://onw20Mining%20&%20Exportation%20of%20Limestone.pdf">0n%20Mining%20&%20Exportation%20of%20Limestone.pdf</a>
- 223. FOUNDATIONS FOR COMPETITIVENESS AND GROWTH PROJECT Loan

- No.:8408-JM Assignment Title: Jamaica's Limestone Industry V theiGuides Admin, accessed August 31, 2025,
- http://admin.theiguides.org/Media/Documents/FCG-CON-049\_Limestone%20TO R%20-FINAL.pdf
- 224. The new Management Plan 2025-2030 Yorkshire Dales National Park, accessed August 31, 2025,
  - https://www.yorkshiredales.org.uk/about/national-park-management-plan/partnership/the-new-management-plan-2025-2030/
- 225. Nullarbor bioregion DCCEEW, accessed August 31, 2025, <a href="https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833">https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833</a> <a href="https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833">https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833</a> <a href="https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833">https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833</a> <a href="https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833">https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833</a> <a href="https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833">https://www.dcceew.gov.au/sites/default/files/env/resources/a8015c25-4aa2-4833</a> <a href="https://www.dcceew.gov.au/sites/env/resources/a8015c25-4aa2-4833">https://www.dcceew.gov.au/sites/env/resources/a8015c25-4aa2-4833</a> <a href="https://www.dcceew.gov.au/sites/env/resources/env/resources/au/sites/env/resources/au/sites/env/resources/au
- 226. The Nullarbor Anything but Plain Journey Beyond Rail, accessed August 31, 2025.
  - https://www.journeybeyondrail.com.au/news/the-nullarbor-anything-but-plain/
- 227. Nullarbor Plains Xeric Shrublands One Earth, accessed August 31, 2025, <a href="https://www.oneearth.org/ecoregions/nullarbor-plains-xeric-shrublands/">https://www.oneearth.org/ecoregions/nullarbor-plains-xeric-shrublands/</a>
- 228. accessed January 1, 1970, https://www.france.fr/en/destination/jura-mountains